

# JGR Solid Earth

# FEATURE ARTICLE

10.1029/2018JB016460



#### **Key Points:**

- Waning volcanic degassing along the southern Eurasian margin is a possible cause of the long-term Cenozoic climate cooling
- A climate change-volcanism feedback during glacial-interglacial cycles explains the change in shape of late Cenozoic climate oscillations

**Correspondence to:** P. Sternai,

pietro.sternai@unimib.it

#### Citation:

Sternai, P., Caricchi, L., Pasquero, C., Garzanti, E., van Hinsbergen, D. J. J., & Castelltort, S. (2020). Magmatic forcing of Cenozoic climate? Journal of Geophysical Research: Solid Earth, 125, e2018JB016460. https://doi.org/ 10.1029/2018JB016460

Received 23 MAR 2019 Accepted 9 NOV 2019 Accepted article online 20 DEC 2019

# **Magmatic Forcing of Cenozoic Climate?**

Pietro Sternai<sup>2</sup>, Luca Caricchi<sup>1</sup>, Claudia Pasquero<sup>2</sup>, Eduardo Garzanti<sup>2</sup>, Douwe J. J. van Hinsbergen<sup>3</sup>, and Sébastien Castelltort<sup>1</sup>

<sup>1</sup>Department of Earth Sciences, University of Geneva, Geneva, Switzerland, <sup>2</sup>Department of Earth and Environmental Sciences, University of Milano-Bicocca, Milan, Italy, <sup>3</sup>Department of Earth Sciences, Utrecht University, Utrecht, Netherlands

**Abstract** Established theories ascribe much of the observed long-term Cenozoic climate cooling to atmospheric carbon consumption by erosion and weathering of tectonically uplifted terrains, but climatic effects due to changes in magmatism and carbon degassing are also involved. At timescales comparable to those of Milankovitch cycles, late Cenozoic building/melting of continental ice sheets, erosion, and sea level changes can affect magmatism, which provides an opportunity to explore possible feedbacks between climate and volcanic changes. Existing data show that extinction of Neo-Tethyan volcanic arcs is largely synchronous with phases of atmospheric carbon reduction, suggesting waning degassing as a possible contribution to climate cooling throughout the early to middle Cenozoic. In addition, the increase in atmospheric  $CO_2$  concentrations during the last deglaciation may be ascribed to enhanced volcanism and carbon emissions due to unloading of active magmatic provinces on continents. The deglacial rise in atmospheric  $CO_2$  points to a mutual feedback between climate and volcanism mediated by the redistribution of surface masses and carbon emissions. This may explain the progression to higher amplitude and increasingly asymmetric cycles of late Cenozoic climate oscillations. Unifying theories relating tectonic, erosional, climatic, *and magmatic* changes across timescales via the carbon cycle offer an opportunity for future research into the coupling between surface and deep Earth processes.

**Plain Language Summary** Among the most fascinating contemporary developments in the Earth Sciences is the idea that plate tectonics and climate changes are coupled through complex cycles. More than four decades of study have revealed tantalizing examples of relationships between processes near the Earth's surface and deeper within. Accounting for such a *surface-deep Earth process coupling* has improved our understanding of major climate and tectonic events for virtually all timescales. So far, however, the role of magmatism was overlooked. Magmatism exerts a strong influence on the amount of greenhouse gasses in the atmosphere because of volcanic outgassing. In turn, tectonic and climatic changes control the transfer of rocks, water, and ice masses at the Earth surface and within the Earth's interior, thereby affecting the production, transfer, and eruption of magmas. Unravelling the interactions between surface and deep Earth processes accounting for magmatism will help managing natural resources and mitigating natural hazards. Even more importantly, understanding how climate is naturally related to plate tectonics and magmatism will allow scientists to assess the anthropogenic perturbations to the Earth system and anticipate ongoing and future climate changes.

# 1. Rationale: Coupling Surface and Deep Earth Processes

Mountain ranges, sedimentary basins, subduction zones, volcanic arcs, oceanic ridges, and nearly all major geological features testify fluxes of geological materials occurring *at and across* the Earth's surface. Interactions between the solid Earth (i.e., our planet's solid surface and its interior) and the fluid Earth (i.e., our planet's wet or icy surface and the atmosphere) are thus ubiquitous and occur across timescales (e.g., Broecker, 2018). Correspondences between global climatic and tectonic changes throughout the Cenozoic fostered our understanding of the evolution of mountain ranges (e.g., Molnar & England, 1990), rock exhumation (Ruddiman, 1997), and the geological carbon cycle (e.g., Raymo & Ruddiman, 1992). The main logical link to relate past climate and tectonic changes is their synchronicity. However, limited temporal resolution in the geological archives prevents a clear recognition of the causative relationships behind such changes. Which mechanisms control the coupling between surface and deep Earth processes? Which are the characteristic timescales and magnitudes of the feedbacks involved? Answering these questions is still a challenge in the Earth Sciences (e.g., NAP, 2012; Frontiers, Media, 2015; Huntington et al.,

©2019. American Geophysical Union. All Rights Reserved. 2017). A continuously growing body of observational constraints and advances in coupled processes modeling increases opportunities to progress in this research.

Here, we focus on the Cenozoic (last ~65 Myr), which includes several well-documented natural experiments reporting on the coupling between solid and fluid Earth processes (e.g., Miller et al., 1987; Zachos et al., 2008). In section 2, we address processes occurring at multimillion years timescales. First, we review an ongoing debate on causal relationships between the India-Eurasia convergence and collision history and climate cooling in the Neogene-Quaternary via erosion and weathering of tectonically uplifted terrains. We then develop the debate supporting the hypothesis that variations in volcanic degassing due to extinction of Neo-Tethyan volcanic arcs may have contributed to climate cooling since the Eocene. In section 3, we address the coupling between surface and deep processes at timescales comparable to those of Milankovitch cycles. Previous studies suggested that the waxing and waning of continental ice sheets, erosion, and sea level changes during climate oscillations affect the volcanic activity. Here, a simple model of radiative equilibrium is used to evaluate whether estimated variations of atmospheric CO<sub>2</sub> concentrations due to volcanic changes induced by the surface mass redistribution throughout glacial-interglacial cycles may affect the amplitude and symmetry of Pleistocene climate oscillations consistently with observations. In section 4, we outline open questions and opportunities for research focused on the coupling between surface and deep Earth processes. Quantifying feedbacks between climate and tectonics accounting for magmatism is a compelling challenge to progress our understanding of the coupling between surface and deep Earth processes, toward a holistic understanding of the functioning of the Earth system.

# 2. Solid Earth Control on Cenozoic Climate Cooling

The oxygen isotope composition of ocean sediments is sensitive to ocean temperature and ice volumes and shows unsteady but continuous increase in  $\delta^{18}$ O values since ~50 Ma (Figure 1a). This implies long-term global cooling following the late Paleocene-early Eocene (~60–50 Ma) climate warming (e.g., Zachos et al., 2001). Antarctica was ice free until about ~30 Ma (e.g., Barrett et al., 1987; DeConto & Pollard, 2003; Pagani et al., 2011), and glaciation of the Northern Hemisphere only started during the Plio-Quaternary (last ~5 Myr; e.g., Raymo, 1994; Willeit et al., 2015). Thus, early Cenozoic  $\delta^{18}$ O trends are driven primarily by temperature changes. The variability of the  $\delta^{18}$ O record increased after ~30 Ma and became even larger in the last ~3 Myr, suggesting an increasingly dominant coupling with ice cover (e.g., Zachos et al., 2001; see section 3). Enhanced differentiation of the midlatitude continents into areas of wetter and drier climates is also inferred from the terrestrial fossil record and abundant glacial detritus blanketing middle-to-high latitudes worldwide (e.g., Ruddiman et al., 1989). For more than a century, scientists have been searching for explanations for the Cenozoic climate evolution (e.g., Broecker, 2018; Chamberlin, 1899), often alluding to a dominant, but still elusive, role of plate tectonics.

#### 2.1. Tectonic Influence on the Ocean and Atmosphere Circulation Patterns

Paleogeographic changes, involving the formation and destruction of oceanic corridors and orographic barriers, are a straightforward mechanism through which plate tectonics may affect climate. Overall, global temperatures decrease when increasing portions of energy-absorbing oceans at the tropics are replaced by energy-reflective continental land (e.g., Crowley et al., 1987). However, continental migration toward the tropics over the past 100 Myr is limited and unlikely to explain Cenozoic cooling (e.g., Barron, 1985; Goddéris et al., 2014). Thermal insulation due to the development of the circum-Antarctic currents, when Antarctica separated from Australia in the middle to late Oligocene (e.g., Barrett et al., 1987; Kennett et al., 1974), has also been invoked to explain synchronous cooling of Antarctica (e.g., DeConto & Pollard, 2003; Pagani et al., 2011). However, models suggest that this event would also reduce precipitation in the polar region, in turn inhibiting the onset of glaciation (e.g., Oglesby, 1989). A long-standing idea is that the emergence of the Isthmus of Panama in the late Pliocene may have triggered the onset of Northern Hemisphere glaciations (e.g., Bartoli et al., 2005; Haug & Tiedemann, 1998; Raymo, 1994). However, numerical simulations suggest that high-latitude cooling may also occur without the closure of the Panama gateway (e.g., Maier-Reimer et al., 1990; Murdock et al., 1997). An additional hypothesis is that Cenozoic true polar wander-the rotation of the Earth and mantle relative to the spin axis in response to change in solid Earth density distributions-may have preconditioned the northern Atlantic for glaciation (Steinberger et al., 2015).





**Figure 1.** Compilation of datasets reporting on Cenozoic trends. (a)  $\delta^{18}$ O, surface temperature, and the atmospheric CO<sub>2</sub> concentration required to yield the global temperature change if climate sensitivity is 0.75 °C/W/m<sup>2</sup> (Hansen et al., 2013; Zachos et al., 2001). Orange dots show the CO<sub>2</sub> proxy records as from Beerling and Royer (2011) and references therein (error bars are not shown for figure legibility). PETM, EECO and MECO stand for Paleocene-Eocoene Thermal Maximum, Early-Eocene Climate Optimum and Middle-Eocene Climate Optimum, respectively. (b) India-Asia convergence (van Hinsbergen et al., 2011). (c) Mean Tibetan Plateau elevation; solid line from Fielding (1996), dashed line from Xu (1981), and diamond from DeCelles et al. (2007). Pre-Oligocene values (dotted line) are speculative (see text). (d) Global sediment yield (Peizhen et al., 2001). (e) Magmatic activity (nonexhaustive list; see also section 2.2.2 and references therein).

Climatic effects have also been linked to late Cenozoic growth of the Tibetan Plateau, the greatest topographic feature on Earth, which is inferred to have perturbed atmospheric circulation at the scale of the entire Northern Hemisphere (e.g., Ruddiman & Raymo, 1988; Ruddiman & Kutzbach, 1989; Molnar et al., 2010). Topography in the Tibetan Plateau was built up in several steps, likely starting well before the collision between India and Asia (Murphy et al., 1997; Wang et al., 2017). Paleoelevations may have been comparable to the present since at least the late Oligocene (DeCelles et al., 2007; Dupont-Nivet et al., 2008; Van Der Beek et al., 2009), but estimated plateau elevations during the Eocene vary from near sea level to ~5 km (e.g., Botsyun et al., 2019; Ding et al., 2014; Wei et al., 2016). Normal faulting in Tibet at about 13 Ma suggests an increase in Plateau elevation at that time (Molnar et al., 1993; Murphy et al., 2009) and may have had climatic effects such as initiation of aeolian accumulation in the main part of the Chinese Loess Plateau and the onset of intense monsoonal circulation (e.g., Sun et al., 2008, 2014). Numerical models indicate that a lowaltitude Tibetan Plateau before the Eocene would have resulted in a trend from equable moist temperate





Figure 2. Schematic representation of sources and sinks of surface carbon.

climates in the early Cenozoic to increased seasonality and regional differentiation of climate in the Northern Hemisphere today (Ruddiman & Kutzbach, 1989), overall consistent with paleobotanical and paleoenvironmental records (e.g., Hren et al., 2009; Xu, 1981). However, predictions do not include a pronounced drop in high-latitude temperatures as the plateau elevations are increased, even when feedbacks (e.g., snow albedo, sea ice, and mixed-layer ocean temperatures) are taken into account. The development of high topography and/or the formation of marine sills or gateways alone thus seems insufficient to drive the long-term Cenozoic evolution. Additional factors such as changes in atmospheric or ocean composition are required (e.g., Raymo & Ruddiman, 1992).

#### 2.2. Tectonic Influence on the Geological Carbon Cycle

At timescales of millions to tens of millions of years, the geological carbon cycle modulates the storage of carbon into rocks and the release of carbon into the ocean and atmosphere (e.g., Sundquist, 1985), thereby linking the evolution of climate and life to plate tectonics (Figure 2). Variations in the ocean and atmosphere concentration of radiative gases (particularly  $CO_2$ ) can alter the Earth's climate. These express a dynamic equilibrium between sinks of carbon by chemical weathering of silicate minerals, precipitation/deposition of carbonates in the ocean and burial of organic matter in sediments (e.g., Galy et al., 2007), and sources of carbon by volcanic emissions, hydrothermal systems, and metamorphism (e.g., Walker et al., 1981; Berner & Lasaga, 1989; Berner, 2003). Relevant reactions can be simplified as

$$CaSiO_3 + CO_2 CaCO_3 + SiO_2$$

Geological proxies suggest decreasing atmospheric  $CO_2$  levels since the Triassic followed by an early Cenozoic (~60–50 Ma) increase in  $CO_2$  values (Figure 1a), including the Paleocene-Eocene Thermal Maximum at ~55.5 Ma and the Early-Eocene Climate Optimum at 52.9–50.7 Ma (e.g., Anagnostou et al., 2016; Beerling & Royer, 2011; Hansen et al., 2013; Zachos et al., 2001). Then, two subsequent phases of reduction of atmospheric  $CO_2$  concentrations occurred. The first phase lasted until the Oligocene and included ~500 kyr of climate amelioration, the middle Eocene Climate Optimum, at ~40 Ma (e.g., Sluijs et al., 2013). The second phase began in the middle Miocene and continues today (although it is currently outpaced by short-term natural and anthropogenic emissions). Two end-member hypotheses have been formulated to explain the long-term Cenozoic cooling via a tectonic-controlled reduction in atmospheric carbon levels: Climate cooling may be driven either primarily by an increase in the global weathering rate (the main sink of surface carbon) on the one hand and by a decrease in the global degassing rate (the main source of surface carbon) on the other hand. We review these two hypotheses hereafter.

#### 2.2.1. Increase in Global Weathering

Chemical weathering is a function of continental relief (e.g., Raymo et al., 1988; Ruddiman, 1997; Von Blanckenburg, 2006). During the Cenozoic, the convergence between India and Eurasia resulted in the deformation, uplift, and erosion of the Indian passive margin in the Himalayas, and in the formation of

the Tibetan Plateau in Southern Asia. In this region, incident solar heating in summer drives the strong atmospheric convection and rainfall associated with the Asian monsoon (e.g., Molnar et al., 1993). The presence of a large elevated plateau in the proximity of a warm ocean leads to very high weathering rates in the Himalayan region (Pinet & Souriau, 1988). Time constraints on the growth of the Tibetan Plateau to presentday elevations, however, are ambiguous (e.g., Botsyun et al., 2019; van Hinsbergen & Boschman, 2019; Figures 1b and 1c and section 2.1), which hampers establishing a straightforward correlation between climate cooling and the reduction of atmospheric  $CO_2$  concentration due to Himalayan-Tibetan uplift and associated enhanced weathering. As an alternative or complementary mechanism, Jagoutz et al. (2016) propose atmospheric  $CO_2$  drawdown due to weathering of mafic and ultramafic rocks between ~50 and 40 Ma (see also Macdonald et al., 2019).

The shift toward a cooler and stormier climate may have affected erosion also through a change from fluvialto glacial-dominated conditions since the Plio-Quaternary (e.g., Molnar & England, 1990). The possible fourfold to fivefold increase in sediment deposition rates in most marine and oceanic basins since ~5 Ma (Figure 1d) would suggest increased erosion of most orogenic belts worldwide, ascribed to the onset of the Northern Hemisphere glaciation (e.g., Hay et al., 1988; Donnelly, 1982; Peizhen et al., 2001). Molnar and England (1990) further proposed that enhanced erosion and valley carving in the Plio-Quaternary would explain inferred synchronous uplift of mountain peaks worldwide (e.g., in the Alps, Pyrenees, and Tibet) through isostatic rebound of orogens following unloading. This idea provoked ongoing controversies because a global increase in mountain erosion should increase global weathering and CO<sub>2</sub> consumption (e.g., Riebe et al., 2004) and thus be the cause rather than the effect of climate cooling. On the one hand, the global increase in erosion and sedimentation may be related to a preservation bias in the sedimentary record (i.e., "Sadler effect"; Willenbring & von Blanckenburg, 2010; Willenbring & Jerolmack, 2016). On the other hand, the thermochronological record, which should be insensitive to preservation biases, corroborates the worldwide acceleration of mountain erosion (Herman et al., 2013). Although the analysis by Herman et al. (2013) was recently questioned (Schildgen et al., 2018), a number of geochemical proxies (e.g., Li and Sr isotopes) suggest a change in ocean composition during the late Cenozoic that may be related to increased worldwide erosion and weathering (e.g., Broecker, 2018; Edmond, 1992; Misra & Froelich, 2012). Assessing the links between mountain uplift, erosion, and weathering has been a major concern of research regarding the surface-deep Earth processes coupling. However, it remains unclear whether tectonics controls climate through mountain uplift, erosion, and weathering or rather if climate forces mountain uplift through erosion and isostatic rebound during the late Cenozoic. Despite a lack of consensus, more than four decades of intense debate suggest that a mutual feedback between tectonic uplift of mountain ranges and climate cooling via erosion and weathering does exist. Alternatively, or in addition, changes of global degassing rate following major tectonic events contributed to Cenozoic climate changes (e.g., Johnston et al., 2011; McKenzie et al., 2016; Brune et al., 2017; Godderis & Donnadieu, 2017), a possibility that is examined in more detail below.

#### 2.2.2. Decrease in Global Degassing

Although climate cooling since the Triassic has been shown to correlate with changes in plate boundary magmatism (e.g., Li & Elderfield, 2013; Van Der Meer et al., 2014), the decrease in global volcanic degassing hypothesis received less attention compared to the increase in global weathering model of Cenozoic climate cooling. The overall temporal correspondence between waning of Neo-Tethyan volcanism and major climate deterioration throughout the Eocene and early Oligocene suggests a direct causal relationship between these observations. Yet, as we shall outline hereafter, other major geodynamic events occurred at roughly the same time and may have conditioned volcanic outgassing and the Cenozoic climate history. In addition, the extinction of volcanic arcs is inherently related to orogenic processes, which involve changes in erosion and weathering. Thus, the contributions of magmatic and weathering changes to the Cenozoic climate evolution are related. Our purpose here is to point out the potential climatic implications of magmatic changes throughout the Cenozoic, while constraining both contributions of magmatism and weathering to Cenozoic climate changes should be the objective of future research.

#### 2.2.2.1. Neo-Tethys Closure and Associated Magmatism

Throughout the Mesozoic and into the Eocene, the Neo-Tethyan seaways extended east-west for more than 15,000 km (e.g., Seton et al., 2012). In the Late Cretaceous, the Neo-Tethys Ocean included a composite, east-west trending subduction system about 13,000 km long, with associated volcanic arcs along the





Figure 3. Paleotectonic maps of the Neo-Tethyan margin (modified after the paleotectonic reconstructions by the DARIUS program, 2018, http://istep.dgs.jussieu. fr/darius/maps.html).

southern continental margin of Eurasia and within the Neotethys ocean (e.g., Dercourt et al., 2000; Stampfli & Borel, 2002; Maffione et al., 2017; Guilmette et al., 2018). A double-subduction system is unequivocal between Arabia and Eurasia (e.g., Agard et al., 2007; van Hinsbergen, Maffione, et al., 2019) and was also proposed to have existed between India and Asia (e.g., Aitchison et al., 2007; Jagoutz et al., 2015), although the evidence there is debated (e.g., Hu et al., 2016; Huang et al., 2015). In any case, subduction of Indian plate lithosphere remained active until well after initial collision of the northernmost continental crust of the Indian plate with Asia (Figure 3). The timing of initial India-Asia collision is also controversial, with proposed ages from >60 and <35 Ma (e.g., Molnar & Tapponnier, 1975; Searle et al., 1987; Garzanti et al.,



1987; Aitchison et al., 2007; Bouilhol et al., 2013; Orme et al., 2015; Garzanti, 2008), partly depending on the assumed definition of continental collision (e.g., the end of marine sedimentation versus the end of oceancontinent subduction) or the method used to constrain the age of collision (e.g., stratigraphy, sedimentology, and style of deformation). The arrival of the edge of the Indian continental margin at the trench adjacent to southern Tibet (Beck et al., 1996) has been recently constrained as  $59 \pm 1$  Ma (Decelles et al., 2014; Wu et al., 2014; Hu et al., 2015).

The climax of arc magmatism, however, trailed initial collision by some 7 Myr and is followed by progressive demise. Evidence of extensive Late Cretaceous to Paleogene volcanic activity is widespread along the southern margin of Eurasia (Figure 3). In southern Tibet, the Gangdese volcanic arc was voluminous until early Eocene times (e.g., Dewey et al., 1988; Treolar et al., 1996; Yin & Harrison, 2000) and recorded an ignimbrite flare-up stage around 52 Ma (e.g., Ji et al., 2009; Zhu et al., 2015), followed by return to background activity and slow extinction lasting until the late Eocene (e.g., Sanchez et al., 2013). The short-lived magmatic flareup is in striking coincidence with the Early-Eocene Climate Optimum, suggesting a cause-effect relationship between the magmatic and climatic event, a possibility that was only marginally investigated thus far. Subduction of the carbonate-rich Indian passive margin succession in the middle Paleocene may have enhanced carbon recycling at the Transhimalayan volcanic arc (e.g., Kent & Muttoni, 2008; van Hinsbergen et al., 2019; Hoareau et al., 2015). Hoareau et al. (2015) modeled the volume of subducted materials and the amount of CO<sub>2</sub> emitted along the northern Tethyan margin and the effects of estimated CO<sub>2</sub> fluxes on global climate. Enhanced CO<sub>2</sub> degassing due to subduction of thicker Indian continental margin sediments after ~60 Ma would raise surface temperatures to some extent, consistently with the observed early Eocene climate warming. Similarly, Kerrick and Caldeira (1994) suggested that carbon degassing due to metamorphism of Indian margin rocks along the Himalayan belt enhanced paleoatmospheric carbon levels, although revised estimates point to a minor contribution to early Eocene warming (Kerrick & Caldeira, 1998). A reduction of carbon dioxide emissions is inferred between ~50 and 40 Ma, synchronous with the observed phase of climate cooling, when Tibetan arc volcanism was waning after collision between India and Eurasia (Hoareau et al., 2015; Jagoutz et al., 2016; Figure 1).

Between ~50 and ~35 Ma there has been the demise of a highly productive magmatic arc/back-arc system across much of Iran and adjacent areas (e.g., Agard et al., 2011; Schleiffarth et al., 2018). This magmatism was a net source of atmospheric CO<sub>2</sub>. Peak magmatism along the Arabia-Eurasia margin between ~45 and 40 Ma (e.g., Kazmin et al., 1986; Alpaslan et al., 2004; Arslan & Aslan, 2006; Vincent et al., 2005; Figure 1e) could have promoted a rise in atmospheric carbon concentrations during the middle Eocene climate optimum (Allen & Armstrong, 2008), which was previously attributed to an unspecified rise in igneous activity (Bohaty & Zachos, 2003). Only limited volcanic activity persisted into the Oligocene in Iran, though minor and sporadic magmatic events continue until the present (e.g., Pearce et al., 1990). Waning arc magmatism during the late Eocene in Iran and Anatolia is consistent with synchronous compressional deformation and topographic uplift (e.g., Allen & Armstrong, 2008; Gürer & van Hinsbergen, 2019; Vincent et al., 2005). The reduction in arc magmatism would have diminished carbon degassing into the atmosphere, and consequently global temperatures, consistently with observations (Figures 1e and 3).

#### 2.2.2.2. Examples of Other Major Geodynamic Events and Associated Magmatic Changes

Obviously, this is not to dismiss the possible contribution of magmatic activity from igneous provinces other than the Southern Eurasian margin to early or middle Cenozoic climate changes. For instance, propagation of the Atlantic mid-ocean spreading between Eurasia and Greenland along the Reykjanes Ridge occurred by ~58 Ma (e.g., Seton et al., 2012; Figure 4) and the associated release of carbon likely contributed to the early Eocene climate amelioration. In the Caribbean region, there was continental underthrusting and ophiolite emplacement between 50 and 40 Ma on Cuba and Hispaniola (e.g., Boschman et al., 2014; Iturralde-Vinent et al., 2008), which may have contributed to waning global outgassing. Starting at ~39 Ma, large volumes of magmas were erupted from several volcanic centers in the Basin and Range, where magmatism ended by ~33 Ma (Gans, 1989), in remarkable synchronicity with the final phases of latest Eocene climate cooling. Ophiolite emplacement on Kamchatka and the Kuriles and the subsequent onset of subduction along the Aleutian and the Kuril trenches following arc-continent collision in Kamchatka and, farther south, along the Izu-Bonin-Marianas trench over a length comparable to that of the Neo-Tethyan margin occurred around ~50 Ma (e.g., Konstantinovskaia, 2001; Ishizuka et al., 2011; Domeier et al., 2017; Vaes et al., 2019; Figure 4). However, intraoceanic subduction zones commonly involve small amounts of carbon emissions





**Figure 4.** Global plate reconstruction at ~60 and ~40 Ma (modified after Seton et al., 2012). Blue and red lines show major subduction zones or thrusts and oceanic ridges or transform, respectively. The green line in the upper panel shows the position of India as suggested from the sedimentary and stratigraphic record (e.g., Hu et al., 2015), instead of the seafloor spreading history and global plate motion reconstruction (Seton et al., 2012). Yellow arrows point to major modifications of plate boundaries involving potentially significant magmatic changes: Demise of the Neo-Tethyan volcanic arc, onset of the Izu-Bonin-Marianas subduction, and propagation of ocean spreading between Greenland and Eurasia. Black arrows show the direction of motion of tectonic plates.

because few carbonate sediments are subducted (e.g., Johnston et al., 2011), either due to the distance from the continents or because intraoceanic trenches are typically deeper than the carbonate compensation depth. Tertiary volcanism in the Great Basin of western United States began in the Eocene with predominantly effusive and explosive volcanism and ended with late Oligocene and Miocene (~26–15 Ma) ignimbrite eruptions (Gans, 1987). Similarly, increased convergence rate normal to the Chilean Andes at ~25 Ma was followed by broadening of the volcanic arc into adjacent portions of Bolivia and Argentina (Pilger, 1984). The number of volcanic ash layers and volumetric estimates of igneous rocks erupted in the circum-Pacific region suggest increased volcanism in the Miocene (e.g., Donnelly, 1973; Kennett et al., 1977). These events may well have contributed to early and middle Miocene climate warming through enhanced carbon degassing (Kashiwagi & Shikazono, 2003). Although a number of previous global analyses show clear temporal correlations between volcanism and climate changes in the Paleozoic and Mesozoic (e.g., Lee et al.,





**Figure 5.** Estimated surface temperature change using the GEOCARB III model (http://climatemodels.uchicago.edu/geocarb/) imposing a decrease in global CO<sub>2</sub> degassing rate by 1% (dashed line) and 10% (solid line) from a spin-up CO<sub>2</sub> degassing rate of  $10 \times 10^{12}$  mol/a. The reader is referred to Berner and Kothavala (2001), specially Appendix 1, and Archer et al. (2009) for the relevant equations in GEOCARB III. Imposed boundary conditions are shown on the plot.

2013; McKenzie et al., 2016; Van Der Meer et al., 2014), these correlations have been barely tested for the entire Cenozoic.

#### 2.3. Modeling Climatic Effects of Neo-Tethyan Arc Extinction

A global-scale analysis is required to assess thoroughly the potential magmatic control of Cenozoic climate changes. However, we emphasize here the role played by progressive extinction of Neo-Tethyan volcanic arcs during the Eocene (i.e., until ~35 Ma; Figures 1 and 3) and of the consequent reduction in global carbon emissions, which must have contributed at least to some extent to long-term Cenozoic climate cooling. To evaluate the climatic effects of the extinction of Neo-Tethyan volcanic arcs, it is necessary to translate fluxes of erupted magma into carbon fluxes. We have calibrated this relationship using the Indonesian arc as a modern analogue of the early Cenozoic Neo-Tethyan arc, because (i) it lies along the same plate margin, (ii) it shared a similar pre-Cenozoic and early Cenozoic tectonic and paleogeographic evolution, and (iii) it measures about the same length as the eastern Neo-Tethyan margin (pre-India-Eurasia collision) and western Neo-Tethyan margin (pre-Arabia-Eurasia collision; e.g., Ricou, 1994; Hall, 2002).

According to Burton et al. (2013, and references therein), estimated modern carbon emissions from the Indonesian volcanic arc contribute to global natural volcanic emissions by 1–10%. If these reference values are used as forcing terms for GEOCARB III models of the atmospheric  $pCO_2$  and temperature evolution (Archer et al., 2009; Berner & Kothavala, 2001) and all other variables are kept constant, then predictions show cooling trends consistent in magnitude and timescale with that observed between, for example, 50 and 45 Ma (Figures 1 and 5). These first-order estimates are consistent with previous modeling efforts (e.g., Kashiwagi &

Shikazono, 2003) and suggest that a causal link between extinction of the Neo-Tethyan volcanic arcs and climate cooling throughout the Eocene-Oligocene is plausible.

We hope that these considerations may stimulate more thorough and systematic global-scale research efforts focused at constraining the relationships between magmatism and climate changes throughout the Cenozoic. We particularly stress that even though orogenic processes and the arrest of arc volcanism associated with continental subduction/collision are considered to produce effects on climate through enhanced erosion/weathering and reduced outgassing, respectively, the relationships between continental subduction/collision and magmatic activity are to date particularly poorly constrained.

# 3. Mutual Feedback Between Quaternary Climate and Volcanism

If climate changes are related to variations of the magmatic activity at long and intermediate timescales, then climate and magmatism may be even more tightly linked at short timescales such as those of Milankovitch cycles. Over the late Pliocene and Quaternary, benthic  $\delta^{18}$ O values of deep-sea sediments and the Antarctic ice core records (see upper panels in Figure 6) show global temperature and ice volume oscillations, both expressions of glacial-interglacial cycles (e.g., Lisiecky & Raymo, 2007). Variations of insolation (i.e., solar energy received on the Earth's surface) and atmospheric CO<sub>2</sub> concentrations are primary drivers of climate oscillations and associated glacial-interglacial cycles (e.g., Bereiter et al., 2015; Imbrie, 1984). These show a puzzling progressive increase in amplitude and shift toward longer cooling than warming trends over the late Pleistocene (e.g., Lisiecky & Raymo, 2007). Because the parameters of the Earth's orbit vary within relatively stable boundaries at systematic Milankovitch periodicities and the solar radiation is constant at Myr timescales, it is safe to assume that the insolation curve maintained its amplitude and symmetry throughout at least the Quaternary (e.g., Broecker & Donk, 1970). Therefore, the change in shape of late Pleistocene climate oscillations must be driven by dynamics internal to the Earth system (e.g., Ashkenazy & Tziperman, 2004; Huybers, 2007; Liu & Herbert, 2004; Raymo, 1992). We hereby evaluate the possibility that such dynamics involve a feedback loop between glacial-interglacial cycles and volcanism via the



**Figure 6.** (a) Global  $\delta^{18}$ O (Zachos et al., 2001) and temperature (Hansen et al., 2013) curves and tephra frequency in the Northwest Pacific drill (Prueher & Rea, 2001) over the last 5 Ma. The histogram is a three-point moving average after binning into 100 ka slots (see also Kutterolf et al., 2019). Periods of increased volcanic activity around the Pacific Ocean are shown in the lower panel (Kennett & Thunell, 1975; Kennett et al., 1977; Hein et al., 1978; Cambray & Cadet, 1994; Shane et al., 1995; Carey & Sigurdsson, 2000). (b) Global  $\delta^{18}$ O curve (Zachos et al., 2001) and tephra frequency at the Izu Bonin Mariana arc (central panel, using 10 ka binning after Schindlbeck et al., 2018) and the Pacific Ring of Fire (lower panel, using 1 ka binning after Kutterolf et al., 2013). The Middle Pleistocene Transition leads from dominant ~40 ka periodicity of climate oscillations to dominant ~100 ka cycles. Vertical gray bars mark marine isotope stages (MIS) after Railsback et al. (2015).

building/melting of ice sheets, erosion, and volcanic  $CO_2$  degassing that affected the shape of late Pleistocene climate oscillations, as previously hypothesized by, for example, Huybers and Langmuir (2009).

#### 3.1. Glacial-Interglacial Cycles and the Forcing on Volcanism

In the past few decades, it has been proposed that climate changes may affect volcanic activity (Kutterolf et al., 2019, and references therein). Studies of several glaciated regions worldwide (e.g., Sigvaldason et al., 1992; MacLennan et al., 2002; Nowell et al., 2006; Rawson et al., 2015; Praetories et al., 2016) showed a significant increase in subaerial volcanic activity during postglacial warming. Huybers and Langmuir (2009) analyzed a global record of subaerial volcanic eruptions back to the last glacial period and showed an increase in eruption frequency of at least 50% during deglaciation. This correspondence is classically explained by a deglacial-triggering hypothesis (Hall, 1982), according to which the removal of ice caps during interglacials and associated continental lithospheric unloading leads to increased subaerial volcanic activity by reducing the confining pressure within crustal magma chambers (e.g., Sigvaldason et al., 1992; Jellinek et al., 2004; Pagli & Sigmundsson, 2008) and enhancing mantle melting (e.g., Jull & McKenzie, 1996; Tuffen, 2010). The modeling of Sternai et al. (2016) further indicates that erosion changes during glacial-interglacial cycles may produce similar unloading effects on magmatism as the melting of continental ice sheets.

Several authors identified relationships between volcanic cycles and orbital forcing at ~23, ~41, and ~100 kyr (Milankovitch, cycles). Paterne et al. (1990) recognized five periods of enhanced volcanic activity of the

100



Campanian province (Italy) with an ~23 kyr periodicity, correlating also with compositional changes in volcanic products. McGuire et al. (1997) observed enhanced tephra accumulation in the Mediterranean every ~22 kyr throughout the last ~80 kyr, correlating with the rate of sea level change. Glazner et al. (1999) investigated the distribution of intraplate volcanic events in eastern California over the last 800 kyr and found a correlation with glacial maxima. Jellinek et al. (2004) found an ~40 kyr periodicity (and more ambiguous ~17 and 23 kyr periodicities) in the first 400 kyr of the same data set as that of Glazner et al. (1999) and showed that periods of fast ice volume reduction anticipate volcanic pulses by ~3 and ~11 kyr for silicic and basaltic magmas, respectively. Schindlbeck et al. (2018) analyzed the ~1.1 Myr long tephra record of the Izu-Bonin arc (within the IODP Hole U1437B) and showed statistically significant spectral peaks with ~100 kyr periodicity (Figure 6), occurring systematically a few thousands of years after glacial maxima over the last ~0.7 Myr. Kutterolf et al. (2013) compiled a time history of volcanism from around the Pacific Ring of Fire, which accounts for about half of the global length of active plate subduction, and recognized an ~41 kyr periodicity, a correlation with the highest rate of decreasing global ice volumes, and a time lag between deglaciation and volcanic peaks of ~4 kyr. Further large-scale magmatic effects are observed in response to major modifications of surface conditions. Cambray et al. (1995) and Prueher and Rea (1998, 2001) emphasize the synchronism between the onset of the Northern Hemisphere glaciation at ~2.5 Ma and intense volcanic activity all around the North Pacific Ocean possibly due to sea level lowering (Figure 6), a link that was already hypothesized by Kennett and Thunell (1975). Sternai et al. (2017) proposed a link between sea level drop during the Messinian salinity crisis and a pulse of magmatic events in Mediterranean igneous provinces.

Melting ice sheets on land has the downstream consequence of raising sea level. This would imply that submarine volcanism may be inhibited by a deeper water column at the same time as subaerial volcanic eruptions are enhanced by glacial unloading on land. The seafloor bathymetry, primarily determined by oceanic crustal thickness, suggests a sensitivity of mid-ocean ridge volcanism to sea level changes on Milankovitch timescales (e.g., Crowley et al., 2015; Lund & Asimow, 2011; Schindlbeck et al., 2018; Tolstoy, 2015). A peak of hydrothermal activity ~15 ka after the last glacial maximum on the Mid-Atlantic Ridge (Middletone et al., 2016), Juan de Fuca Ridge (Costa et al., 2017), and at the East Pacific Rise (Lund et al., 2016) has been interpreted as the delayed melt formation following the hydrostatic pressure minimum.

#### 3.2. Volcanic Forcing on the Shape of Glacial-Interglacial Cycles

The solid Earth vents  $CO_2$  into the atmosphere through volcanism and magmatic processes, which appear to vary in response to the redistribution of the surface ice, water, and rock masses during glacial-interglacial cycles. Because changes in  $CO_2$  removal rate by chemical weathering have a negligible effect on timescales comparable to those of glacial-interglacial cycles (e.g., Le Hir et al., 2009; Colbourn et al., 2015), the following question arises: Can variations of global  $CO_2$  volcanic degassing driven by glacial-interglacial cycles affect the shape of climate oscillations? Since glaciation and deglaciation produce anticorrelated variations in subaerial and submarine volcanism (see section 3.1), the relative magnitude of these effects has to be quantified to answer the question.

Assuming that emissions are proportional to the frequency of eruptions, Huybers and Langmuir (2009) estimate the global evolution of volcanic CO<sub>2</sub> emissions since 40 ka (Figure 7a). Their model indicates an atmospheric  $CO_2$  reduction by 5–20 ppm during the last glacial period (~40–18 ka), marginally consistent with the observed ~20 ppm decrease in atmospheric CO<sub>2</sub> data from the Dome C and Taylor Dome Antarctic ice cores (Indermühle et al., 2000; Monnin et al., 2004). During the initial phase of deglaciation (~18-13 ka) their model suggests a 5-40 ppm increase in CO<sub>2</sub> concentration, whereas observations show an ~50 ppm increase. During advanced deglaciation (~13-7 ka), the reconstruction indicates a 15-70 ppm increase in volcanic CO2, consistent with the observed ~40 ppm increase. According to their analysis, equilibration with oceanic volcanism compensates for up to  $\sim 20\%$  of the increase in subaerial volcanic CO<sub>2</sub> flux during the deglaciation. The overall consistency between estimated volcanic emissions by Huybers and Langmuir (2009) and observed atmospheric concentrations of CO<sub>2</sub> suggests that volcanic changes may indeed be primary drivers of the glacialinterglacial climate variability. Huybers and Langmuir (2009) construct their time history of CO<sub>2</sub> fluxes using eruption frequencies rather than unloading histories, and so the potential contribution of erosion to increased volcanic degassing during the deglaciation (Sternai et al., 2016) is already accounted for. However, the contribution of erosion changes to unloading of active magmatic provinces is not acknowledged in the classic deglacial-triggering hypothesis of volcanic events (e.g., Hall, 1982; Sigvaldason et al., 1992; Jull &





**Figure 7.** (a) CO<sub>2</sub> concentrations from the Dome C and Taylor Dome Antarctic ice cores 4 (dots), and the data smoothed over a 2 ka window (solid line) follow the left axis. Modeling results by Huybers and Langmuir (2009) constraining the contribution to atmospheric CO<sub>2</sub> from volcanic activity (dashed line) follow the right axis. The dark and light gray shaded regions represent the 90% confidence interval on the modeling estimates and the interglacial period, respectively. (b) Boxes represent ranges of erosion rates from glaciated catchments or proximal basins including errors in estimations (vertical) and resolved timescale (horizontal; Brandon et al., 1988; Hallet et al., 1996; Sheaf et al., 2003; Reiners et al., 2003; Koppes et al., 2006; Hebbeln et al., 2007; Geirsdottir et al., 2007; Berger & Spotila, 2008; Koppes et al., 2009; Cowton et al., 2012; Herman et al., 2013; Herman & Brandon, 2015). AL, Alaska; SA, Southern Andes; GR, Greenland; NO, Norway; PNW, Pacific Northwest; IC, Iceland. The vertical dashed line represents the approximate time since the LGM (Lisiecky & Raymo, 2007). (c) Estimated continental unloading rate owing to constant melting of 1,000 m of ice (dotted line for reference) and erosion (assuming a surface rock density of 2,700 kg/m<sup>3</sup>) throughout the interglacial. (d) Equilibrium change in global mean surface air temperature resulting from increasing atmospheric CO<sub>2</sub> concentrations (see text for detail).

McKenzie, 1996; Jellinek et al., 2004; Pagli & Sigmundsson, 2008; Tuffen, 2010). It thus seems important to remark here that given the density contrast between ice and surface rocks, erosion rate changes by  $\sim$ 1 mm/a or higher during deglaciation unload continents by a similar or greater amount than the melting of ice sheets (Sternai et al., 2016). Long-term proxies of global sediment efflux from mountainous regions show just such erosion rate variability (e.g., Peizhen et al., 2001; Koppes & Montgomery, 2009; Herman et al., 2013; Figures 7b and 7c). At secular to millennial timescales, erosional fluxes may be even higher (e.g., Koppes & Montgomery, 2009) and subject to strong variations due to modifications of the subglacial hydraulic system and water supply (e.g., Andrews et al., 2014; Flowers & Clarke, 2002; Herman et al., 2011; Sternai et al., 2013). The abrupt and high-magnitude magmatic pulses involved by such erosional changes are likely to force centennial- to millennial-scale variations of atmospheric greenhouse gases seemingly unrelated to ocean dynamics (e.g., Marcott et al., 2014). In addition, while reduced submarine volcanism due to sea level rise during interglacials buffers increased subaerial CO<sub>2</sub> emissions, the erosional forcing on subaerial emissions is unbalanced because the downstream effect of erosion is depositional loading of epicontinental or abyssal marine basins mostly located away from oceanic ridges.



It thus seems plausible that enhanced subaerial volcanic degassing due to combined ice melting and erosion may account for all of the observed ~90 ppm increase in atmospheric  $CO_2$  concentration during the deglaciation, that is, an increase of ~50% with respect to glacial concentrations (Figures 7a and 7d). Arrhenius law may be used to calculate the resulting radiative forcing, rf, so that  $rf = \alpha \cdot \ln \frac{C}{C_0}$  where  $\alpha$  is a constant and  $\frac{C}{C_0}$  is the relative increase in CO<sub>2</sub> concentration. The equilibrium change in global mean near-surface air temperature,  $T_s$ , is then calculated as  $T_s = \lambda \cdot rf$ , where  $\lambda$  is the climate sensitivity. Assuming  $\lambda$  equal to 0.8 K/W/m<sup>2</sup> (e.g., Royer et al., 2007), a  $T_s$  of about 2 °C solely due to volcanic changes throughout glacial-interglacial cycles should be expected (Figure 7d). This volcanic forcing of temperature oscillations is concordant and largely in phase with that related to variations of the Earth's albedo during formation and melting of continental ice caps, producing an additional gain of surface temperatures through glacialinterglacial cycles of comparable magnitudes (e.g., Broccoli & Manabe, 1987; Dickinson et al., 1987; Meehl & Washington, 1990). A gain of temperature oscillations by up to ~4 °C during the late Pleistocene with respect to prior the onset of the Northern Hemisphere glaciation is consistent with observations (Hansen et al., 2013; see top panel in Figure 6a). An increase in the amplitude of temperature oscillations is also observed after the Eocene-Oligocene boundary, when the Antarctic ice sheet is established (Figure 1a). The mutual influence between volcanism and climate oscillations in the presence of continental ice sheets, amplified by albedo effects and larger continental ice volumes in the Pleistocene, may thus represent a primary driver of the late Cenozoic progression toward higher-amplitude climate oscillations.

The climate-volcanism feedback during glacial-interglacial cycles would also generate the asymmetric cooling and warming trends of late Pleistocene climate oscillations (e.g., Lisiecky & Raymo, 2007). The main logic is that inhibition of subaerial eruptions during glacial periods forces accumulation of gasses in magmatic reservoirs, which are then released over a few thousands of years during the early interglacials (e.g., Jellinek et al., 2004). Assuming that the long-term weathering  $CO_2$  sink is at equilibrium with the steady state volcanic CO<sub>2</sub> outgassing, the weathering carbon sink slightly dominates over inhibited volcanic carbon emissions when ice sheets grow, leading to a temporary reduction of atmospheric CO<sub>2</sub>, which sustains climate cooling. As soon as the orbital forcing of solar radiation overtakes the threshold to trigger the deglaciation, enhanced volcanic carbon outgassing dominates over the weathering carbon sink, in turn fostering climate warming and bringing the overall atmospheric CO<sub>2</sub> budget back to equilibrium. If, for instance, enhanced outgassing is extinguished in ~10 ka, then the cooling phase is forced to ~30 ka for early Pleistocene 40 ka cycles and to ~90 ka for late Pleistocene 100 ka cycles. The ~10 ka time window is chosen arbitrarily, but the duration of warming of late Pleistocene climate oscillations and the expected response time of magmatic systems to surface load changes constrain this value (e.g., Jellinek et al., 2004; Lisiecky & Raymo, 2007). Because both constraints are largely independent on the period of climate oscillations (i.e., 40 or 100 ka), fixing the duration of the phase of enhanced outgassing allows us to speculate that the longer the period of climate oscillations, the longer the phase of gas accumulation in magmatic reservoirs, the larger the outgassing rate during the phase of enhanced emissions, and the more pronounced the asymmetry between warming and cooling trends (Figure 8). This is consistent with the increasing amplitude and asymmetry of climate oscillations throughout the mid-Pleistocene transition from dominant 40 to 100 ka cycles (e.g., Lisiecky & Raymo, 2007). We hope that this statement may trigger and motivate further research on the volcano-climatic feedback during glacial-interglacial cycles.

# 4. Surface-Deep Earth Processes Coupling: Challenges and Opportunities

Once plate tectonics theory became established (e.g., Le Pichon, 1968; McKenzie & Parker, 1967), the correspondences between major geodynamic, magmatic, and climatic changes were readily recognized (e.g., Kennett & Thunell, 1975; Kennett et al., 1977; Ziegler et al., 1979; Raymo & Ruddiman, 1992; McKenzie et al., 2016). Assessing the physical mechanisms and chemical reactions that allow for the surface-deep Earth processes coupling has been a major concern of research ever since. Scientists focused on processes such as mountain building, erosion, and the waxing and waning of continental ice sheets. Disassembling a complex system of interactions into simpler relationships has been the main approach to progress on this research, and a broad range of geological data sets were used to develop theories relating mountain building and erosion, and next erosion and atmospheric  $CO_2$  consumption. However, at the multimillion years timescale, besides erosion and weathering of uplifted terrains, mountain building following continental



**Figure 8.** Thick lines show modeled atmospheric CO<sub>2</sub> variability associated with normalized sinusoidal radiative forcing at 40 kyr (thin blue line) and 100 kyr (thin red line). Shaded bars indicate the phase of enhanced outgassing during the deglaciation lasting for ~10 kyr (see text for detail). Ocean and atmospheric CO<sub>2</sub> are assumed to be in equilibrium, as their equilibration timescale is much shorter than the timescales considered here (Archer, 2005). Weathering is the only CO<sub>2</sub> sink considered and the weaterhing rate,  $\dot{W}$ , is assumed to be constant and equal to 0.1 Gt/a (Gaillardet et al., 1999; Moon et al., 2014). The volcanic CO<sub>2</sub> emission rate is assigned a glacial value,  $\dot{e}$ , and an enhanced value,  $\dot{E}$ , during the deglaciation calculated as  $\dot{E}=2\dot{e}$  (Huybers & Langmuir, 2009). We also assume that the CO<sub>2</sub> budget is balanced over an entire climate cycle, so that  $\dot{W}(g+d) = \dot{E}d+\dot{e}g$ , where g and d are the duration of the glacial and deglacial periods, respectively. Thus, for 100 kyr climate cycles with deglaciation lasting for 10 kyr (see text for detail),  $\dot{e}=0.09$  Gt/a and  $\dot{E}=0.13$  Gt/a. For 40 kyr climate cycles with deglaciation lasting for 10 kyr, we adopt  $\dot{e}=0.09$  Gt/a, leading to  $\dot{E}=0.13$  Gt/a.

collision/subduction involves the closure of oceanic seaways and is typically followed, after a certain time lapse, by extinction of volcanic arcs, which also impact on climate. Early Cenozoic climate cooling is widely ascribed to atmospheric carbon consumption due to enhanced erosion and weathering following initiation of Himalayan growth, but Paleocene closure of Neo-Tethys and final waning of arc magmatism are currently unconstrained factors. In addition, the ~10 Myr long phase of climate warming preceding the onset of the Cenozoic cooling trend cannot be readily explained by theories referring only to enhanced weathering following topographic uplift because enhanced weathering involves a reduction of atmospheric CO<sub>2</sub> and thus climate cooling. Similarly, known effects of surface processes on magmatism and of magmatism on climate have not been duly considered to assess whether enhanced erosion following mountain building led to the onset of glaciation in the late Cenozoic or the onset of glaciation enhanced erosion rates and lithospheric unloading, thereby increasing mountain uplift through isostatic adjustment. Until the magmatic forcing of Cenozoic climate is quantified, the contribution of erosion/weathering of uplifted terrains to climate cooling will remain elusive. Continental collision/subduction and associated orogenesis increase weathering and cause arc demise, and both processes work in tandem to reduce the atmospheric carbon budget. Discerning between the two contributions to the Cenozoic climate evolution as indicated by paleoclimate proxies is a major challenge for future research. The links between surface mass redistribution and volcanism characterized by recent research focused on Plio-Quaternary glacial-interglacial cycles seem to offer an appropriate starting point for the development of a unifying theory relating tectonics, surface processes, magmatic, and climatic changes. In this way, we may eventually capture the essence of the surface-deep Earth processes coupling and thus gain new insights into the functioning of the Earth system (Figure 9).

The development and validation of such a theory involves a range of disciplines of the Earth Sciences and will require the conjugation and systematic analyses of global data and modeling. Besides extending the body of observational constraints reporting on the Cenozoic tectonic and sedimentary fluxes, improving data sets on the timing, magnitude, and chemical composition of major magmatic events is needed. Interpreting available data will require implementation of current models to account for phase transitions and/or multiphase flow in order to investigate the coevolution of geodynamic and magmatic processes. The continuously





**Figure 9.** Schematic representation of the surface-deep Earth processes coupling. Assessing the role of magmatism in setting Cenozoic climate changes is one of the Grand Challenges for future research on the surface-deep Earth processes coupling.

increasing amount of experimental constraints on rock rheology and petrology and recent improvements in numerical techniques allowed the development of petrothermomechanical numerical models capable of handling intense deformation and magmatism (e.g., Bouilhol et al., 2015; Gerya & Yuen, 2007; Menant et al., 2016; Roche et al., 2018). A component of surface processes is often included into recent geodynamic numerical models (e.g., Ueda et al., 2015) and can benefit from a wide range of continuously growing geomorphological, sedimentological, biostratigraphical, and geophysical observations. Climate models are also continuously improved in order to account for multiple factors pertaining to the solid Earth such as variations of the tectonic or sedimentary fluxes (e.g., Archer et al., 2009; Zeebe, 2012). When boundary conditions cannot be constrained by direct observations, estimates stemming from independent petrothermomechanical and/or surface processes modeling results may be useful. Although challenging, assessing first-order relationships between major tectonic and climatic changes accounting for magmatism seems feasible as of now.

To what extent did the extinction of Neo-Tethyan volcanic arcs and/or other major geodynamic events affect global carbon degassing throughout

the Cenozoic? Would associated changes in surface CO<sub>2</sub> concentrations involve a climatic response consistent with paleoclimate proxies? Understanding the post-50 Ma climate cooling may first require understanding the ~60-50 Ma climate warming, seemingly unrelated to weathering changes and suggesting a prominent magmatic forcing on climate. To this aim, we particularly stress the need for accurate and systematic reconstructions of plate kinematics and magmatic activity at a global scale to be used as observational constraints for modeling the associated perturbations of the geological carbon cycles. To what extent can the variability of atmospheric carbon during glacial-interglacial cycles be ascribed to changes in volcanic activity? Do glaciation and associated redistribution of surface masses act as phase locker between volcanic and climatic changes through a modulation of carbon emissions? Do anticorrelated variations of submarine and subaerial volcanism throughout glacial-interglacial cycles add up to positive or negative perturbations of the global long-term carbon degassing and how would these perturbations modulate the shape of Late Quaternary climate oscillations? Understanding the relationships between Plio-Quaternary climate oscillations and volcanism requires accurate quantifications of the surface ice water and rock mass redistribution associated with glaciation and in-depth assessments of the response of magmatic systems to perturbations of the surface loads. In general, any contribution that improves our understanding of carbon recycling at plate margins, volcanic arc extinction, surface load variations during glacial-interglacial cycles, and the carbon emission budget of submarine and subaerial volcanism will help answering these and other outstanding questions regarding the coupling between surface-Earth and deep-Earth processes.

Assessing the climatic response to long- and short-term magmatic forcing throughout the Cenozoic will not only be of scientific relevance but also produce invaluable societal implications. For instance, early Cenozoic climate changes provide insights into the coupling of climate and the carbon cycle and thus may help to predict the consequences of current carbon emissions in the future (e.g., Zachos et al., 2008). In addition, glaciers and ice sheets are a primary resource of fresh water and are often used in critical ways, while geothermal power plants around volcanic contexts worldwide should produce about 140 GWe (~8.3% of total electricity production) by 2050 (e.g., Bertani, 2016). Because enforced deglaciation since the industrial era may lead to increased volcanic activity and hazard (e.g., Pagli & Sigmundsson, 2008; Tuffen, 2010), understanding the natural joint variability of glaciated and volcanic environments is an urgent action to undertake in order to improve our ability to manage natural resources and hazards in fast-evolving and high-potential/risk terrains.

# 5. Boxed Text

The geological carbon cycle regulates the storage of  $CO_2$  into rocks and the release of  $CO_2$  into the ocean and atmosphere via weathering of surface minerals and volcanic emissions, thereby linking the evolution of



climate and life to plate tectonics. The last ~60 Myr are characterized by a long-term climate cooling trend, which may be ascribed to a combination of reduced atmospheric  $CO_2$  due to increased global weathering and/or progressive reduction of  $CO_2$  emissions following extinction of volcanic arcs. Assessing the magmatic contribution to the long-term climate evolution is particularly challenging because our knowledge regarding volcanic carbon emission and the timing and modalities of volcanic arc generation and extinction is still flawed. Major research efforts are thus required and should involve global-scale analyses of available data and the development of new observational constraints and models.

Climate cooling throughout the Cenozoic Era led to the onset of Northern Hemisphere glaciation in the Plio-Quaternary (~5 Ma to present). Because volcanic emissions affect climate, but the redistribution of surface ice, water, and rock masses associated with glacial-interglacial cycles affects volcanic activity, a feedback loop between volcanic and climatic change does exist. Paleoclimate proxies show a progressive increase of the amplitude and asymmetry of climate oscillations throughout the Late Quaternary, and our preliminary analysis suggests that the mutual feedback between volcanism and climate during icehouse stages may explain such observations. However, more thorough quantifications of the surface mass redistribution during glacial-interglacial cycles and a better understanding of the sensitivity of volcanic systems to surface load variations are required to assess the variability of subaerial and submarine volcanic carbon degassing and validate this proposal.

By supporting the view that the Cenozoic climate evolution is largely controlled by changes in magmatic activity, we put forth a major challenge for future research, because assessing the magmatic forcing on climate requires global-scale, multidisciplinary investigations. Improving data and modeling regarding the strain, topographic and magmatic evolution of plate boundaries, and the redistribution of surface mass across the Earth's surface caused by glacial-interglacial cycles and the associated volcanic response are specifically required to tackle this challenge. Understanding past climate changes will allow us to better assess ongoing modifications and predict future conditions, which in turn should influence policymakers toward responsible, conscious, and shared decisions for adaptation and mitigation options. However, the challenge does not only involve scientists but all individuals that contribute sensitizing the community to the complex network of interactions that control the Earth's climate. This is an essential step toward increasing awareness of the societal impact on our planet.

#### Acknowledgments

Pietro Sternai was supported by the Swiss NSF (Ambizione Grant PZ00P2 168113/1) and the Italian MIUR (Rita Levi Montalcini Grant, DM 694-26/2017). This study is also part of the project MIUR-Dipartimenti di Eccellenza 2018-2022, Department of Earth and Environmental Sciences. University of Milano-Bicocca. Laurent Jolivet, Marco Malusà, and Andrea Zanchi are thanked for helpful suggestions. The manuscript was significantly improved based on revisions by Guilhem Hoareau and an anonymous reviewer. The data supporting this research are obtained from published articles cited in the text. Newly proposed models, included in Figures 3, 5, and 7 and explained in the text, are archived on and freely available from MyGeoHub (https:// mygeohub.org/resources/draft/1462? step=5) to comply with FAIR Data standards (https://copdess.org/ enabling-fair-data-project/).

### References

- Agard, P., Jolivet, L., Vrielynck, B., Burov, E., & Monié, P. (2007). Plate acceleration: The obduction trigger? *Earth and Planetary Science Letters*, 258, 428–441.
- Agard, P., Omrani, J., Jolivet, L., Whitechurch, H., Vrielynck, B., Spakman, W., et al. (2011). Zagros orogeny: A subduction-dominated process. *Geological Magazine*, 148(5-6), 692–725.
- Aitchison, J. C., Ali, J. R., & Davis, A. M. (2007). When and where did India and Asia collide? *Journal of Geophysical Research*, 112, B05423. https://doi.org/10.1029/2006JB004706
- Allen, M., & Armstrong, H. A. (2008). Arabia–Eurasia collision and the forcing of mid-Cenozoic global cooling. *Palaeogeography, Palaeoclimatology, Palaeoecology, 265,* 52–58.
- Alpaslan, M., Frei, R., Boztug, D., Kurt, M. A., & Temel, A. (2004). Geochemical and Pb-Sr-Nd isotopic constraints indicating an enrichedmantle source for late Cretaceous to early Tertiary volcanism, Central Anatolia, Turkey. International Geology Review, 46, 1022–1041.
- Anagnostou, E., John, E. H., Edgar, K. M., Foster, G. L., Ridgwell, A., Inglis, G. N., et al. (2016). Changing atmospheric CO<sub>2</sub> concentration was the primary driver of early Cenozoic climate. *Nature*, *533*(7603), 380.
- Andrews, L. C., Catania, G. A., Hoffman, M. J., Gulley, J. D., Luthi, M. P., Ryser, C., et al. (2014). Direct observations of evolving subglacial drainage beneath the Greenland Ice Sheet. *Nature*, 514, 80–83.
- Archer, D. (2005). Fate of fossil fuel CO<sub>2</sub> in geologic time. Journal of Geophysical Research, 110, C09S05. https://doi.org/10.1029/ 2004JC002625
- Archer, D., Eby, M., Brovkin, V., Ridgwell, A., Cao, L., Mikolajewicz, U., et al. (2009). Atmospheric lifetime of fossil fuel carbon dioxide. Annual Review of Earth and Planetary Sciences. https://doi.org/10.1146/annurev.earth.031208.100206
- Arslan, A., & Aslan, Z. (2006). Mineralogy, petrography and whole-rock geochemistry of the Tertiary granitic intrusions in the Eastern Pontides, Turkey. *Journal of Asian Earth Sciences*, *27*, 177–193.

Ashkenazy, Y., & Tziperman, E. (2004). Are the 41 kyr glacial oscillations a linear response to Milankovitch forcing? *Quaternary Science Reviews*, 23(18-19), 1879–1890.

- Barrett, P., Elston, D. P., Harwood, D. W., McKelvey, B. C., & Webb, P. N. (1987). Mid-Cenozoic record of glaciation and sea-level change on the margin of the Victoria Land basin, Antarctica. *Geology*, 15(7), 634–637.
- Barron, E. J. (1985). Explanations of the Tertiary global cooling trend. Palaeogeography, Palaeoclimatology, Palaeoeclogy, 50(1), 45–61.
  Bartoli, G., Sarnthein, M., Weinelt, M., Erlenkeuser, H., Garbe-Schönberg, D., & Lea, D. W. (2005). Final closure of Panama and the onset of Northern Hemisphere glaciation. *Earth and Planetary Science Letters*, 237(1), 33–44.
- Beck, R. A., Burbank, D. W., Sercombe, W. J., Khan, A. M., & Lawrence, R. D. (1996). Late Cretaceous ophiolite obduction and Paleocene India-Asia collision in the westernmost Himalaya. *Geodinamica Acta*, 9(2-3), 114–144.

Beerling, D. J., & Royer, D. L. (2011). Convergent cenozoic CO2 history. Nature Geoscience, 4(7), 418.



- Bereiter, B., Eggleston, S., Schmitt, J., Nehrbass-Ahles, C., Stocker, T. F., Fischer, H., et al. (2015). Revision of the EPICA Dome C CO<sub>2</sub> record from 800 to 600 kyr before present. *Geophysical Research Letters*, *42*, 542–549. https://doi.org/10.1002/2014GL061957
- Berger, A. L., & Spotila, J. A. (2008). Denudation and deformation in a glaciated orogenic wedge: The St. Elias orogen, Alaska. *Geology*, 36, 523–526.
- Berner, R. A. (2003). The long-term carbon cycle, fossil fuels and atmospheric composition. Nature, 426(6964), 323.
- Berner, R. A., & Kothavala, Z. (2001). GEOCARB III: A revised model of atmospheric CO<sub>2</sub> over Phanerozoic time. American Journal of Science, 304, 397–437.
- Berner, R. A., & Lasaga, A. C. (1989). Modeling the geochemical carbon cycle. Scientific American, 260(3), 74-81.
- Bertani, R. (2016). Geothermal power generation in the world 2010–2014 update report. Geothermics, 60, 31–43.
- Bohaty, S. M., & Zachos, J. C. (2003). Significant Southern Ocean warming event in the late middle Eocene. Geology, 31, 1017–1020.
- Boschman, L. M., van Hinsbergen, D. J. J., Torsvik, T. H., Spakman, W., & Pindell, J. L. (2014). Kinematic reconstruction of the Caribbean region since the Early Jurassic. *Earth-Science Reviews*, *138*, 102–136.
- Botsyun, S., Sepulchre, P., Donnadieu, Y., Risi, C., Licht, A., & Rugenstein, J. K. C. (2019). Revised paleoaltimetry data show low Tibetan Plateau elevation during the Eocene. *Science*, *363*(6430), eaaq1436.
- Bouilhol, P., Jagoutz, O., Hanchar, J. M., & Dudas, F. O. (2013). Dating the India–Eurasia collision through arc magmatic records. Earth and Planetary Science Letters, 366, 163–175.
- Bouilhol, P., Magni, V., van Hunen, J., & Kaislaniemi, L. (2015). A numerical approach to melting in warm subduction zones. *Earth and Planetary Science Letters*, 411, 37–43.
- Brandon, M. T., Roden-Tice, M. K., & Garver, J. I. (1988). Late Cenozoic exhumation of the Cascadia accretionary wedge in the Olympic Mountains, northwest Washington State. *Geological Society of America Bulletin*, 110, 985–1009.
- Broccoli, A. J., & Manabe, S. (1987). The influence of continental ice, atmospheric CO<sub>2</sub>, and land albedo on the climate of the Last Glacial Maximum. *Climate Dynamics*, 1(2), 87–99.
- Broecker, W. (2018). Collision changes everything. Geochemical Perspectives, 7(2), 142-153.
- Broecker, W. S., & Donk, J. (1970). Insolation changes, ice volumes, and the O18 record in deep-sea cores. *Reviews of Geophysics*, 8(1), 169–198.
- Brune, S., Williams, S. E., & Müller, R. D. (2017). Potential links between continental rifting, CO<sub>2</sub> degassing and climate change through time. Nature Geoscience, 10(12), 941.
- Burton, M. R., Sawyer, G. M., & Granieri, D. (2013). Deep carbon emissions from volcanoes. *Reviews in Mineralogy and Geochemistry*, 75, 323–354.
- Cambray, H., & Cadet, J. (1994). Testing global synchronism in peri-Pacific arc volcanism. Journal of Volcanology and Geothermal Research, 63(3-4), 145–164.
- Cambray, H., Pubellier, M., Jolivet, L., & Pouclet, A. (1995). Volcanic activity recorded in deep-sea sediments and the geodynamic evolution of western Pacific island arcs. In B. Taylor, & J. Natland (Eds.), Active margins and marginal basins of the western Pacific, Geophysical monograph, (pp. 97–124). Washington: American Geophysical Union.
- Carey, S., & Sigurdsson, H. (2000). Grain size of Miocene volcanic ash layers from Sites 998, 999, and 1000: Implications for source areas and dispersal. In R. M. Leckie, H. Sigurdsson, G. D. Acton, & G. Draper (Eds.), *Proceedings ODP, Scientific Results* (Vol. 165, pp. 101–110). College Station, TX: Ocean Drilling Program.
- Chamberlin, T. C. (1899). An attempt to frame a working hypothesis of the cause of glacial periods on an atmospheric basis. *The Journal of Geology*, 7(6), 545–584.
- Colbourn, G. A., Ridgwell, A., & Lenton, T. M. (2015). The time scale of the silicate weathering negative feedback on atmospheric CO<sub>2</sub>. *Global Biogeochem Cycles*, 29, 583–596. https://doi.org/10.1002/2014GB005054
- Costa, K. M., McManus, J. F., Middletone, J. L., Langmuir, C. H., Huybers, P. J., Winckler, G., & Mukhopadhyay, S. (2017). Hydrothermal deposition on the Juan de Fuca Ridge over multiple glacialeinterglacial cycles. *Earth and Planetary Science Letters*, 479, 120–132.
- Cowton, T., Nienow, P., Bartholomew, I., Sole, A., & Mair, D. (2012). Rapid erosion beneath the Greenland ice sheet. Geology, 40, 343-346.
- Crowley, J. W., Katz, R. F., Huybers, P., Langmuir, C., & Park, S.-H. (2015). Glacial cycles drive variations in the production of oceanic crust. *Science*, 347(6227), 1237–1240.
- Crowley, T. J., Mengel, J. G., & Short, D. A. (1987). Gondwanaland's seasonal cycle. Nature, 329(6142), 803.
- Decelles, P. G., Kapp, P., Gehrels, G. E., & Ding, L. (2014). Paleocene-Eocene foreland basin evolution in the Himalaya of southern Tibet and Nepal: Implications for the age of initial India-Asia collision. *Tectonics*, 33, 824–849. https://doi.org/10.1002/2014TC003522
- DeCelles, P. G., Quade, J., Kapp, P., Fan, M., Dettman, D. L., & Ding, L. (2007). High and dry in central Tibet during the Late Oligocene. *Earth and Planetary Science Letters*, 253(3-4), 389–401.
- DeConto, R. M., & Pollard, D. (2003). Rapid Cenozoic glaciation of Antarctica induced by declining atmospheric CO<sub>2</sub>. Nature, 421(6920), 245.
- Dercourt, J., Gaetani, M., Vrielynck, B., Barrier, E., Biju-Duval, B., Brunet, M. F., et al. (2000). Peri-Tethys Palaeogeographical Atlas 2000. Université Pierre et Marie Curie, Paris.
- Dewey, J. F., Shackleton, R. M., Chengfa, C., & Yiyin, S. (1988). The tectonic evolution of the Tibetan Plateau. Philosophical Transactions of the Royal Society A, 327(1594), 379–413.
- Dickinson, R. E., Meehl, G. A., & Washington, W. M. (1987). Ice-albedo feedback in a CO<sub>2</sub>-doubling simulation. *Climatic Change*, 10(3), 241–248.
- Ding, L. Q., Xu, Q., Yue, Y., Wang, H., Cai, F., & Li, S. (2014). The Andean-type Gangdese Mountains: Paleoelevation record from the Paleocene–Eocene Linzhou Basin. *Earth and Planetary Science Letters*, 392, 250–264.

Domeier, M., Shephard, G. E., Jakob, J., Gaina, C., Doubrovine, P. V., & Torsvik, T. H. (2017). Intraoceanic subduction spanned the Pacific in the Late Cretaceous–Paleocene. *Science Advances*, *3*(11), eaao2303.

- Donnelly, T. W. (1982). Worldwide continental denudation and climatic deterioration during the late Tertiary: Evidence from deep-sea sediments. *Geology*, *10*(9), 451–454.
- Dupont-Nivet, G., Hoorn, C., & Konert, M. (2008). Tibetan uplift prior to the Eocene-Oligocene climate transition: Evidence from pollen analysis of the Xining Basin. Geology, 36, 987–990.
- Edmond, J. M. (1992). Himalayan tectonics, weathering processes, and the strontium isotope record in marine limestones. *Science*, 258(5088), 1594–1597.

Donnelly, T. W. (1973). Circum-Caribbean explosive volcanic activity evidence from Leg 15 sediments. In N. T. Edgar, J. B. Saunders, et al. (Eds.), *Initial Reports of the Deep Sea Drilling Project*, (Vol. 15, pp. 969–987).

Fielding, E. J. (1996). Tibet uplift and erosion. Tectonophysics, 260(1-3), 55-84.

Flowers, G., & Clarke, G. (2002). A multicomponent coupled model of glacier hydrology: 1. Theory and synthetic examples. Journal of Geophysical Research, 107(B11), 2287. https://doi.org/10.1029/2001JB001122

Frontiers, Media (2015). Grand challenges in the field of Earth Science. Lausanne: Edited by Collaborative. https://doi.org/10.3389/978-2-88919-711-8

Gaillardet, J., Dupré, B., Louvat, P., & Allegre, C. J. (1999). Global silicate weathering and CO<sub>2</sub> consumption rates deduced from the chemistry of large rivers. *Chemical Geology*, 159(1-4), 3–30.

Galy, V., France-Lanord, C., Beyssac, O., Faure, P., Kudrass, H., & Palhol, F. (2007). Efficient organic carbon burial in the Bengal fan sustained by the Himalayan erosional system. *Nature*, 450, 407–410.

Gans, P. B. (1987). An open-system, two-layer crustal stretching model for the eastern Great Basin. Tectonics, 6(1), 1-12.

Gans, P. B. (1989). Synextensional magmatism in the Basin and Range province: A case study from the eastern Great Basin. Geological Society of America, 233.

Garzanti, E. (2008). Comment on "When and where did India and Asia collide?" by Jonathan C. Aitchison, Jason R. Ali, and Aileen M. Davis. Journal of Geophysical Research, 113, B04411. https://doi.org/10.1029/2007JB005276

Garzanti, E., Baud, A., & Mascle, G. (1987). Sedimentary record of the northward flight of India and its collision with Eurasia (Ladakh Himalaya, India). *Geodinamica Acta*, 1(4-5), 297–312.

Geirsdottir, A., Miller, G. H., & Andrews, J. T. (2007). Glaciation, erosion, and landscape evolution of Iceland. Journal of Geodynamics, 43, 170–186.

Gerya, T. V., & Yuen, D. A. (2007). Robust characteristics method for modelling multiphase visco-elasto-plastic thermo-mechanical problems. *Physics of the Earth and Planetary Interiors*, 163, 83–105.

Glazner, A. F., Manley, C. R., Marron, J. S., & Rojstaczer, S. (1999). Fire or ice: Anti-correlation of volcanism and glaciation in California over the past 800,000 years. *Geophysical Research Letters*, 26, 1759–1762.

Godderis, Y., & Donnadieu, Y. (2017). A sink-or a source-driven carbon cycle at the geological timescale? Relative importance of palaeogeography versus solid Earth degassing rate in the Phanerozoic climatic evolution. *Geological Magazine*, 1–11.

Goddéris, Y., Donnadieu, Y., Le Hir, G., Lefebvre, V., & Nardin, E. (2014). The role of palaeogeography in the Phanerozoic history of atmospheric CO<sub>2</sub> and climate. *Earth-Science Reviews*, *128*, 122–138.

Guilmette, C., Smit, M. A., van Hinsbergen, D. J. J., Gürer, D., Corfu, F., Charette, B., et al. (2018). Forced subduction initiation recorded in the sole and crust of the Semail Ophiolite of Oman. *Nature Geoscience*, *11*(9), 688.

Gürer, D., & van Hinsbergen, D. J. J. (2019). Diachronous demise of the Neotethys Ocean as a driver for non-cylindrical orogenesis in Anatolia. *Tectonophysics*, 760, 95–106.

Hall, K. (1982). Rapid deglaciation as an initiator of volcanic activity: An hypothesis. Earth Surface Processes and Landforms, 7, 45-51.

Hall, R. (2002). Cenozoic geological and plate tectonic evolution of SE Asia and the SW Pacific: Computer-based reconstructions, model and animations. *Journal of Asian Earth Sciences*, 20(4), 353–431.

Hallet, B., Hunter, L., & Bogen, J. (1996). Rates of erosion and sediment evacuation by glaciers: A review of field data and their implications. *Global and Planetary Change*, *12*, 213–235.

Hansen, J., Sato, M., Russell, G., & Kharecha, P. (2013). Climate sensitivity, sea level and atmospheric carbon dioxide. *Philosophical Transactions of the Royal Society A*, 371, 20120294.

Haug, G. H., & Tiedemann, R. (1998). Effect of the formation of the Isthmus of Panama on Atlantic Ocean thermohaline circulation. *Nature*, 393(6686), 673.

Hay, W. W., Sloan, J. L., & Wold, C. N. (1988). Mass/Age distribution and composition of sediments on the ocean floor and the global rate of sediment subduction. Journal of Geophysical Research, 93(B12), 14933–14940.

Hebbeln, D., Lamy, F., Mohtadi, M., & Echtler, H. (2007). Tracing the impact of glacial-interglacial climate variability on erosion of the southern Andes. *Geology*, 35, 131–134. https://doi.org/10.1130/G23243A.1

Hein, J. R., Scholl, D. W., & Miller, J. (1978). Episodes of Aleutian ridge explosive volcanism. Science, 199, 137-141.

Herman, F., Beaud, F., Champagnac, J.-D., Lemieux, J.-M., & Sternai, P. (2011). Glacial hydrology and erosion patterns: A mechanism for carving glacial valleys. *Earth and Planetary Science Letters*, 310, 498–508.

Herman, F., & Brandon, M. (2015). Mid-latitude glacial erosion hotspot related to equatorial shifts in southern Westerlies. *Geology*, 43(11), 987–990.

Herman, F., Seward, D., Valla, P. G., Carter, A., Kohn, B., Willett, S. D., & Ehlers, T. A. (2013). Worldwide acceleration of mountain erosion under a cooling climate. *Nature*, 504.

Hoareau, G., Bomou, B., van Hinsbergen, D. J., Carry, N., Marquer, D., Donnadieu, Y., et al. (2015). Did high Neo-Tethys subduction rates contribute to early Cenozoic warming? *Climate of the Past*, 11(12), 1751–1767.

Hren, M. T., Bookhagen, B., Blisniuk, P. M., Booth, A. L., & Chamberlain, C. P. (2009). 518O and 5D of streamwaters across the Himalaya and Tibetan Plateau: Implications for moisture sources and paleoelevation reconstructions. *Earth and Planetary Science Letters*, 288(1-2), 20–32.

Hu, X., Garzanti, E., Moore, T., & Raffi, I. (2015). Direct stratigraphic dating of India-Asia collision onset at the Selandian (middle Paleocene, 59±1 Ma). *Geology*, 43(10), 859–862.

Hu, X., Garzanti, E., Wang, J., Huang, W., An, W., & Webb, A. (2016). The timing of India-Asia collision onset—Facts, theories, controversies. *Earth-Science Reviews*, 160, 264–299.

Huang, W., van Hinsbergen, D. J. J., Maffione, M., Orme, D. A., Dupont-Nivet, G., Guilmette, C., et al. (2015). Lower Cretaceous Xigaze ophiolites formed in the Gangdese forearc: Evidence from paleomagnetism, sediment provenance, and stratigraphy. *Earth and Planetary Science Letters*, 415, 142–153.

Huntington, K W., Klepeis, K A., (2017). Challenges and opportunities for research in tectonics: Understanding deformation and the processes that link Earth systems, from geologic time to human time. *A community vision document submitted to the U.S. National Science Foundation*.

Huybers, P. (2007). Glacial variability over the last two million years: An extended depth-derived agemodel, continuous obliquity pacing, and the Pleistocene progression. *Quaternary Science Reviews*, 26(1-2), 37–55.

Huybers, P., & Langmuir, C. (2009). Feedback between deglaciation, volcanism, and atmospheric CO<sub>2</sub>. Earth and Planetary Science Letters, 286, 479–491.

Imbrie, J. (1984). Proxy records of Quaternary climate. Annals of Glaciology, 5, 203-203.

Indermühle, A., Monnin, E., Stauffer, B., Stocker, T., & Wahlen, M. (2000). Atmospheric CO<sub>2</sub> concentration from 60 to 20 kyr BP from the Taylor Dome ice core, Antarctica. *Geophysical Research Letters*, 27(5), 735–738.



- Ishizuka, O., Tani, K., Reagan, M. K., Kanayama, K., Umino, S., Harigane, Y., et al. (2011). The timescales of subduction initiation and subsequent evolution of an oceanic island arc. *Earth and Planetary Science Letters*, 306(3-4), 229–240.
- Iturralde-Vinent, M. A., Díaz-Otero, C., García-Casco, A., & van Hinsbergen, D. J. J. (2008). Paleogene foreland basin deposits of North-Central Cuba: A record of arc-continent collision between the Caribbean and North American plates. *International Geology Review*, 50, 863–884.
- Jagoutz, O., Macdonald, F. A., & Royden, L. (2016). Low-latitude arc-continent collision as a driver for global cooling. PNAS, 113(18), 4935–4940.
- Jagoutz, O., Royden, L., Holt, A. F., & Becker, T. W. (2015). Anomalously fast convergence of India and Eurasia caused by double subduction. *Nature Geoscience*, 8, 475–478.
- Jellinek, A. M., Manga, M., & Saar, M. O. (2004). Did melting glaciers cause volcanic eruptions in eastern California? Probing the mechanics of dike formation. *Journal of Geophysical Research*, *109*, B09206. https://doi.org/10.1029/2004JB002978
- Ji, W., Wu, F. Y., Chung, S. L., Li, J., & Liu, C. Z. (2009). Zircon U-Pb geochronology and Hf isotopic constraints on petrogenesis of the Gangdese batholith, southern Tibet. Chemical Geology, 262, 229–245.
- Johnston, F., Turchyn, A. V., & Edmonds, M. (2011). Decarbonation efficiency in subduction zones: Implications for warm Cretaceous climates. *Earth and Planetary Science Letters*, 303, 143–152.
- Jull, M., & McKenzie, D. (1996). The effect of deglaciation on mantle melting beneath Iceland. Journal of Geophysical Research, 101(B10), 21815–21828.
- Kashiwagi, H., & Shikazono, N. (2003). Climate change during Cenozoic inferred from global carbon cycle model including igneous and hydrothermal activities. Palaeogeography, Palaeoclimatology, Palaeoecology, 199, 167–185.
- Kazmin, V. G., Sborshchikov, I. M., Ricou, L. E., Zonenshain, L. P., Boulin, L., & Knipper, A. J. (1986). Volcanic belts as markers of the Mesozoic–Cenozoic active margin of Eurasia. *Tectonophysics*, 123, 123–152.
- Kennett, J. P., Houtz, R. E., Andrews, P. B., Edwards, A. R., Gostin, V. A., Hajos, M., et al. (1974). Development of the circum-Antarctic current. Science, 186(4159), 144–147.
- Kennett, J. P., McBirney, A. R., & Thunell, R. C. (1977). Episodes of Cenozoic volcanism in the circum-Pacific region. Journal of Volcanology and Geothermal Research, 2, 145–163.
- Kennett, J. P., & Thunell, R. C. (1975). Global increase in Quaternary explosive volcanism. Science, 187(4176), 497-502.
- Kent, D. V., & Muttoni, G. (2008). Equatorial convergence of India and early Cenozoic climate trends. Proceedings of the National Academy of Sciences, 105(42), 16065–16070.
- Kerrick, D. M., & Caldeira, K. (1994). Metamorphic CO2 Degassing and Early Cenozoic. GSA Today.
- Kerrick, D. M., & Caldeira, K. (1998). Metamorphic CO<sub>2</sub> degassing from orogenic belts. *Chemical Geology*, 145(3-4), 213–232.
  - Konstantinovskaia, E. A. (2001). Arc-continent collision and subduction reversal in the Cenozoic evolution of the Northwest Pacifc: An example from Kamchatka (NE Russia). *Tectonophysics*, 333, 75–94.
  - Koppes, M, Hallet, B, & Stewart, R. (2006). "Glacial erosion rates and climate influences in Chilean Patagonia." Eos 87, Fall Meet. Suppl., Abstr. C14A-07.
  - Koppes, M. N., Hallet, B., & Anderson, J. A. (2009). Synchronous acceleration of ice loss and glacial erosion, Glaciar Marinelli, Chilean Tierra del Fuego. *Journal of Glaciology*, 55, 207–220.
  - Koppes, M. N., & Montgomery, D. R. (2009). The relative efficacy of fluvial and glacial erosion over modern to orogenic timescales. *Nature Geoscience*, *2*, 644–647.
  - Kutterolf, S., Jegen, M., Mitrovica, J. X., Kwasnitschka, T., Freundt, A., & Huybers, P. (2013). A detection of Milankovitch frequencies in global volcanic activity. *Geology*, 41(2), 227–230.
  - Kutterolf, S., Schindlbeck, J. C., Jegen, M., Freundt, A., & Straub, S. M. (2019). Milankovitch frequencies in tephra records at volcanic arcs: The relation of kyr-scale cyclic variations in volcanism to global climate changes. *Quaternary Science Reviews*, 1–16.
  - Le Hir, G., Donnadieu, Y., Goddéris, Y., Pierrehumbert, R. T., Halverson, G. P., Macouin, M., et al. (2009). The snowball Earth aftermath: Exploring the limits of continental weathering processes. *Earth and Planetary Science Letters*, 277(3-4), 453–463.
  - Le Pichon, X. (1968). Sea-floor spreading and continental drift. Journal of Geophysical Research, 73(12), 3661–3697.
- Lee, C. T. A., Shen, B., Slotnick, B. S., Liao, K., Dickens, G. R., Yokoyama, Y., et al. (2013). Continental arc-island arc fluctuations, growth of crustal carbonates, and long-term climate change. *Geosphere*, 9(1), 21–36.
- Li, G., & Elderfield, H. (2013). Evolution of carbon cycle over the past 100 million years. Geochimica et Cosmochimica Acta, 103, 11-25.
- Lisiecky, L. E., & Raymo, M. E. (2007). Plio–Pleistocene climate evolution: Trends and transitions in glacial cycle dynamics. *Quaternary Science Reviews*, 26, 56–69.
- Liu, Z., & Herbert, T. D. (2004). High-latitude influence on the eastern equatorial Pacific climate in the early Pleistocene epoch. *Nature*, 427(6976), 720.
- Lund, D. C., & Asimow, P. D. (2011). Does sea level influence mid-ocean ridge magmatism on Milankovitch timescales? *Geochemistry, Geophysics, Geosystems, 12*, Q12009. https://doi.org/10.1029/2011GC003693
- Lund, D. C., Asimow, P. D., Farley, K. A., Rooney, T. O., Seeley, E., Jackson, E. W., & Durham, Z. M. (2016). Enhanced East Pacific Rise hydrothermal activity during the last two glacial terminations. *Science*, 351(6272), 478–482.
- Macdonald, F. A., Swanson-Hysell, N. L., Park, Y., Lisiecki, L., & Jagoutz, O. (2019). Arc-continent collisions in the tropics set Earth's climate state. *Science*, *364*, 181–184.
- MacLennan, J., Jull, M., McKenzie, D., Slater, L., & Gronvold, K. (2002). The link between volcanism and deglaciation in Iceland. Geochemistry, Geophysics, Geosystems, 3(11), 1062. https://doi.org/10.1029/2001GC000282
- Maffione, M., van Hinsbergen, D. J. J., de Gelder, G. I. N. O., van der Goes, F. C., & Morris, A. (2017). Kinematics of Late Cretaceous subduction initiation in the Neo-Tethys Ocean reconstructed from ophiolites of Turkey, Cyprus, and Syria. *Journal of Geophysical Research: Solid Earth*, 122, 3953–3976. https://doi.org/10.1002/2016JB013821
- Maier-Reimer, E., Mikolajewicz, U., & Crowley, T. (1990). Ocean general circulation model sensitivity experiment with an open Central American Isthmus. Paleoceanography and Paleoclimatology, 5(3), 349–366.
- Marcott, S. A., Bauska, T. K., Buizert, C., Steig, E. J., Rosen, J. L., Cuffey, K. M., et al. (2014). Centennial-scale changes in the global carbon cycle during the last deglaciation. *Nature*, 514. https://doi.org/10.1038/nature13799
- McGuire, W. J., Howarth, R. J., Firth, C. R., Solow, A. R., Pullen, A. D., Saunders, S. J., et al. (1997). Correlation between rate of sea-level change and frequency of explosive volcanism in the Mediterranean. *Nature*, 389(6650), 473–476. https://doi.org/10.1038/38998 McKenzie, D. P., & Parker, R. L. (1967). The North Pacific: An example of tectonics on a sphere. *Nature*, 216, 1276–1280.
- McKenzie, N. R., Horton, B. K., Loomis, S. E., Stockli, D. F., Planavsky, N. J., & Lee, C. A. (2016). Continental arc volcanism as the principal driver of icehouse-greenhouse variability. *Science*, 352(6284).



- Meehl, G. A., & Washington, W. M. (1990). CO<sub>2</sub> climate sensitivity and snow-sea-ice albedo parameterization in an atmospheric GCM coupled to a mixed-layer ocean model. *Climatic Change*, 16(3), 283–306.
- Menant, A., Sternai, P., Jolivet, L., Guillou-Frottier, L., & Gerya, T. V. (2016). 3D numerical assessments for mantle flow and magma genesis in laterally constrained subduction zones: The eastern Mediterranean case study. *Earth and Planetary Science Letters*, 442, 93–107.
- Middletone, J. L., Langmuir, C. H., Mukhopadhyay, S., McManus, J. F., & Mitrovica, J. X. (2016). Hydrothermal iron flux variability following rapid sea level changes. *Geophysical Research Letters*, 43, 3848–3856. https://doi.org/10.1002/2016GL068408

- Molnar, P., Boos, W. R., & Battisti, D. S. (2010). Orographic controls on climate and paleoclimate of Asia: Thermal and mechanical roles for the Tibetan Plateau. Annual Review of Earth and Planetary Sciences, 38.
- Molnar, P., & England, P. (1990). Late Cenozoic uplift of mountain ranges and global climate change: Chicken or egg? *Nature*, 346(6279), 29–34.
- Molnar, P., England, P., & Martinod, J. (1993). Mantle dynamics, uplift of the Tibetan Plateau, and the Indian monsoon. *Reviews of Geophysics*, 31(4).
- Molnar, P., & Tapponnier, P. (1975). Cenozoic tectonics of Asia: Effects of a continental collision. Science, 189(4201), 419-426.

Monnin, E., Steig, E., Siegenthaler, U., Kawamura, K., Schwander, J., Stauffer, B., et al. (2004). Evidence for substantial accumulation rate variability in Antarctica during the Holocene, through synchronization of CO<sub>2</sub> in the Taylor Dome, Dome C and DML ice cores. *Earth and Planetary Science Letters*, 224(1-2), 45–54.

Moon, S., Chamberlain, C. P., & Hilley, G. E. (2014). New estimates of silicate weathering rates and their uncertainties in global rivers. *Geochimica et Cosmochimica Acta*, 134, 257–274.

- Murdock, T. Q., Weaver, A. J., & Fanning, A. F. (1997). Paleoclimatic response of the closing of the Isthmus of Panama in a coupled oceanatmosphere model. *Geophysical Research Letters*, 24(3), 253–256.
- Murphy, M. A., Saylor, J. E., & Ding, L. (2009). Late Miocene topographic inversion in southwest Tibet based on integrated paleoelevation reconstructions and structural history. *Earth and Planetary Science Letters*, 282(1-4), 1–9.

Murphy, M. A., Yin, A., Harrison, T. M., Durr, S. B., Ryerson, F. J., & Kidd, W. S. F. (1997). Did the Indo-Asian collision alone create the Tibetan Plateau? *Geology*, 25(8), 719–722.

NAP (2012). New research opportunities in the Earth Sciences at the National Science Foundation. Washington, DC: National Research Council.

Nowell, D. A., Jones, M. C., & Pyle, D. M. (2006). Episodic Quaternary volcanism in France and Germany. Journal of Quaternary Science, 21(6), 645–675.

Oglesby, R. J. (1989). A GCM study of Antarctic glaciation. Climate Dynamics, 3(3), 135-156.

Orme, D. A., Carrapa, B., & Kapp, P. (2015). Sedimentology, provenance and geochronology of the upper Cretaceous-lower Eocene western Xigaze forearc basin, southern Tibet. *Basin Research*, 27(4), 387–411.

Pagani, M., Huber, M., Liu, Z., Bohaty, S. M., Henderiks, J., Sijp, W., et al. (2011). The role of carbon dioxide during the onset of Antarctic glaciation. Science, 334(6060), 1261–1264.

- Pagli, C., & Sigmundsson, F. (2008). Will present day glacier retreat increase volcanic activity? Stress induced by recent glacier retreat and its effect on magmatism at the Vatnajo kull ice cap, Iceland. *Geophysical Research Letters*, 35, L09304. https://doi.org/10.1029/ 2008GL033510
- Paterne, M., Labeyrie, J., Guichard, F., Mazaud, A., & Maitre, F. (1990). Fluctuations of the Campanian explosive volcanic activity (South Italy) during the past 190,000 years, as determined by marine tephrochronology. *Earth and Planetary Science Letters*, 98(2), 166–174.

Pearce, J. A., Bender, J. F., Delong, S. F., Kidd, W. S., Low, P. J., Guner, Y., et al. (1990). Genesis of collision volcanism in eastern Anatolia, Turkey. Journal of Volcanology and Geothermal Research, 44, 189–229.

Peizhen, Z., Molnar, P., & Downs, W. R. (2001). Increased sedimentation rates and grain sizes 2–4 Myr ago due to the influence of climate change on erosion rates. *Nature*, 410, 891–897.

Pilger, R. H. (1984). Cenozoic plate kinematics, subduction and magmatism: South American Andes. Journal of the Geological Society, 141(5), 793–802.

Pinet, P., & Souriau, M. (1988). Continental erosion and large-scale relief. Tectonics, 7(3), 563-582.

Praetories, S., Mix, A., Jensen, B., Froese, D., Milne, G., Wolhowe, M., et al. (2016). Interaction between climate, volcanism, and isostatic rebound in Southeast Alaska during the last deglaciation. *Earth and Planetary Science Letters*, 452, 79–89.

Prueher, L. M., & Rea, D. K. (1998). Rapid onset of glacial conditions in the subarctic North Pacific region at 2.67 Ma: Clues to causality. *Geology*, 26, 1027–1030.

- Prueher, L. M., & Rea, D. K. (2001). Tephrachronology of the Kamchatka-Kurile and Aleutian arcs: Evidence for volcanic episodicity. Journal of Volcanology and Geothermal Research, 106, 67–84.
- Railsback, L. B., Gibbard, P. I., Head, M. J., Voarintsoa, N. R., & Toucanne, S. (2015). An optimized scheme of lettered marine isotope substages for the last 1.0 million years, and the climatostratigraphic nature of isotope stages and substages. *Quaternary Science Reviews*, 111, 94–106.

Rawson, H., Naranjo, J. A., Smith, V. C., Fontijn, K., Pyle, D. M., Mather, T. A., & Moreno, H. (2015). The frequency and magnitude of postglacial explosive eruptions at Volcán Mocho-Choshuenco, southern Chile. *Journal of Volcanology and Geothermal Research*, 299, 103–129.

Raymo, M. E. (1992). Global climate change: A three million year perspective. In *Start of a Glacial*, (pp. 207–223). Berlin, Heidelberg: Springer.

Raymo, M. E. (1994). The initiation of Northern Hemisphere glaciation. Annual Review of Earth and Planetary Sciences, 22, 353–383.

Raymo, M. E., & Ruddiman, W. F. (1992). Tectonic forcing of late Cenozoic climate. *Nature*, *359*(6391), 117–122.

Raymo, M. E., Ruddiman, W. F., & Froelich, P. N. (1988). Influence of late Cenozoic mountain building on ocean geochemical cycles. *Geology*, 16, 649–653.

- Reiners, P. W., Ehlers, T. A., Mitchell, S. G., & Montgomery, D. R. (2003). Coupled spatial variations in precipitation and long-term erosion rates across the Washington Cascades. *Nature*, 426, 645–647.
- Ricou, L. E. (1994). Tethys reconstructed-plates, continental fragments and their boundaries since 260-Ma from Central-America to South-Eastern Asia. *Geodinamica Acta*, 7(4), 169–218.

Miller, K. G., Fairbanks, R. G., & Mountain, G. S. (1987). Tertiary oxygen isotope synthesis, sea level history, and continental margin erosion. *Paleoceanography*, 2(1), 1–19.

Misra, S., & Froelich, P. N. (2012). Lithium isotope history of Cenozoic seawater: Changes in silicate weathering and reverse weathering. Science, 335(6070), 818–823.

Riebe, C. S., Kirchner, J. W., & Finkel, R. C. (2004). Erosional and climatic effects in longterm chemical weathering rates in granitic landscapes spanning diverse climate regimes. *Earth and Planetary Science Letters*, 224, 547–562.

Roche, V., Sternai, P., Guillou-Frottier, L., Menant, A., Jolivet, L., Bouchot, V., & Gerya, T. (2018). Emplacement of metamorphic core complexes and associated geothermal systems controlled by slab rollback. *Earth and Planetary Science Letters*, 498, 322–333.

Royer, D., Berner, R., & Park, J. (2007). Climate sensitivity constrained by CO<sub>2</sub> concentrations over the past 420 million years. *Nature*, 446(7135).

Ruddiman, W. F. (1997). Tectonic uplift and climate change. Boston, MA: Springer.

- Ruddiman, W. F., & Kutzbach, J. E. (1989). Forcing of late Cenozoic Northern Hemisphere climate by plateau uplift in southern Asia and the American Wes. *Journal of Geophysical Research*, 94(D15), 18409–18427.
- Ruddiman, W. F., Prell, W. L., & Raymo, M. E. (1989). Late Cenozoic uplift in southern Asia and the American West: Rationale for general circulation modeling experiments. Journal of Geophysical Research, 94(D15), 18379–18391.
- Ruddiman, W. F., & Raymo, M. E. (1988). The past three Millionyears: Evolution of climatic variability in the North Atlantic region, (pp. 227–234). Cambridge: Cambridge University Press.
- Sanchez, V. I., Murphy, M. A., Robinson, A. C., Lapen, T. J., & Heizler, M. T. (2013). Tectonic evolution of the India–Asia suture zone since Middle Eocene time, Lopukangri area, south-central Tibet. Journal of Asian Earth Sciences, 62, 205–220.

Schildgen, T. F., van der Beek, P. A., Sinclair, H. D., & Thiede, R. C. (2018). Spatial correlation bias in late-Cenozoic erosion histories derived from thermochronology. *Nature*, 559(7712), 89.

- Schindlbeck, J. C., Jegen, M., Freundt, A., Kutterolf, S., Straub, S. M., Mleneck-Vautravers, M. J., & McManus, J. (2018). 100-kyr cyclicity in volcanic ash emplacement: Evidence from a 1.1 Myr tephra record from the NW Pacific. *Scientific Reports*. https://doi.org/10.1038/ s41598-018-22595-0
- Schleiffarth, W., Darin, M., Reid, M., & Umhoefer, P. J. (2018). Dynamics of episodic Late Cretaceous–Cenozoic magmatism across Central to Eastern Anatolia: New insights from an extensive geochronology compilation. *Geosphere*, *14*, 1990–2008.
- Searle, M., Windley, B., Coward, M., Cooper, D., Rex, A., Tingdong, L., et al. (1987). The closing of Tethys and the tectonics of the Himalaya. Geological Society of America Bulletin, 98(6), 678–701.

Seton, M., Müller, R. D., Zahirovic, S., Gaina, C., Torsvik, T., Shephard, G., et al. (2012). Global continental and ocean basin reconstructions since 200 Ma. *Earth-Science Reviews*, 113, 212–270.

Shane, P., Froggatt, P., Black, T., & Westgate, J. A. (1995). Chronology of Pliocene and Quaternary bioevents and climatic events from fission-track ages on tephra beds, Wairarapa, New Zealand. *Earth and Planetary Science Letters*, 130, 1241–1254.

Sheaf, M. A., Serpa, L., & Pavlis, T. L. (2003). Exhumation rates in the St. Elias mountains, Alaska. Tectonophysics, 367, 1-11.

Sigvaldason, G. E., Annertz, K., & Nilsson, M. (1992). Effect of glacier loading/deloading on volcanism: Postglacial volcanic production rate of the Dyngjufjiill area, central Iceland. *Volcanology*, 54, 385–392.

Sluijs, A., Zeebe, R. E., Bijl, P. K., & Bohaty, S. M. (2013). A middle Eocene carbon cycle conundrum. Nature Geoscience, 6, 429–434.
Stampfli, G. M., & Borel, G. (2002). A plate tectonic model for the Paleozoic and Mesozoic constrained by dynamic plate boundaries and restored synthetic oceanic isochrons. Earth and Planetary Science Letters, 196, 17–33.

Steinberger, B., Spakman, W., Japsen, P., & Torsvik, T. H. (2015). The key role of global solid-Earth processes in preconditioning Greenland's glaciation since the Pliocene. *Terra Nova*, 27, 1–8.

- Sternai, P., Caricchi, L., Castelltort, S., & Champagnac, J.-D. (2016). Deglaciation and glacial erosion: A joint control
- on magma productivity by continental unloading. *Geophysical Research Letters*, 43, 1632–1641. https://doi.org/10.1002/2015GL067285 Sternai, P., Caricchi, L., Garcia-Castellanos, D., Jolivet, L., Sheldrake, T., & Castelltort, S. (2017). Magmatic pulse driven by sea-level
- changes associated with the Messinian salinity crisis. *Nature Geoscience*. https://doi.org/10.1038/NGEO3032 Sternai, P., Herman, F., Valla, P. G., & Champagnac, J.-D. (2013). Spatial and temporal variations of glacial erosion in the Rhône valley
- (Swiss Alps): Insights from numerical modeling. *Earth and Planetary Science Letters*, *368*, 119–131. Sun, D., Su, R., Bloemendal, J., & Lu, H. (2008). Grain-size and accumulation rate records from Late Cenozoic aeolian sequences in parthern China. Implications for unriting in the East Asian units menagement and unstable transmission and unstable transmission.
- northern China: Implications for variations in the East Asian winter monsoon and westerly atmospheric circulation. *Palaeogeography*, *Palaeoclimatology*, *Palaeoecology*, *264*(1-2), 39–53.
- Sun, J., Xu, Q., Liu, W., Zhang, Z., Xue, L., & Zhao, P. (2014). Palynological evidence for the latest Oligocene–early Miocene paleoelevation estimate in the Lunpola Basin, central Tibet. *Palaeogeography, Palaeoclimatology, Palaeoecology, 99*, 21–30.

Sundquist, E. T. (1985). Geological perspectives on carbon dioxide and the carbon cycle. The carbon cycle and atmospheric CO<sub>2</sub>. In *Natural variations archean to present* (Vol. 32, pp. 55–59).

Tolstoy, M. (2015). Mid-ocean ridge eruptions as a climate valve. *Geophysical Research Letters*, 42, 1346–1351. https://doi.org/10.1002/2014GL063015

- Treolar, P. J., Petterson, M. J., Jan, M. Q., & Sullivan, M. A. (1996). A re-evaluation of the stratigraphy and evolution of the Kohistan arc sequence, Pakistan Himalaya: Implications for magmatic and tectonic arc-building processes. *Journal of the Geological Society, London*, 153, 681–693.
- Tuffen, H. (2010). How will melting of ice affect volcanic hazards in the twenty-first century? *Philosophical Transactions of the Royal Society* A: Mathematical, Physical and Engineering Sciences, 368(1919), 2535–2558.
- Ueda, K., Willett, S. D., Gerya, T., & Ruh, J. (2015). Geomorphological-thermo-mechanical modeling: Application to orogenic wedge dynamics. *Tectonophysics*, 659, 12–30.
- Vaes, B., van Hinsbergen, D. J., & Boschman, L. M. (2019). Reconstruction of subduction and back-arc spreading in the NW Pacific and Aleutian Basin: Clues to causes of Cretaceous and Eocene plate reorganizations. *Tectonics*, 38, 1367–1413. https://doi.org/10.1029/ 2018TC005164
- Van Der Beek, P., Van Melle, J., Guillot, S., Pêcher, A., Reiners, P. W., Nicolescu, S., & Latif, M. (2009). Eocene Tibetan plateau remnants preserved in the northwest Himalaya. *Nature Geoscience*, *2*(5), 364.
- Van Der Meer, D. G., Zeebec, R. E., van Hinsbergen, D. J., Sluijs, A., Spakman, W., & Torsvik, T. H. (2014). Plate tectonic controls on atmospheric CO<sub>2</sub> levels since the Triassic. PNAS, 111(12), 4380–4385.

van Hinsbergen, D. J., & Boschman, L. M. (2019). How high were these mountains? Science, 363, 928–929.

- van Hinsbergen, D. J., Steinberger, B., Doubrovine, P. V., & Gassmöller, R. (2011). Acceleration and deceleration of India-Asia convergence since the Cretaceous: Roles of mantle plumes and continental collision. *Journal of Geophysical Research*, 116, B06101. https://doi.org/ 10.1029/2010JB008051
- van Hinsbergen, D. J. J., Maffione, M., Koornneef, L. M. T., & Guilmette, C. (2019). Kinematic and paleomagnetic restoration of the Semail Ophiolite (Oman) reveals subduction initiation along an ancient Neotethyan fracture zone. *Earth and Planetary Science Letters*. submitted



- van Hinsbergen, D. J. J., Lippert, P. C., Li, S., Huang, W., Advokaat, E. L., & Spakman, W. (2019). Reconstructing Greater India: Paleogeographic, kinematic, and geodynamicperspectives. *Tectonophysics*, *760*, 69–94.
- Vincent, S. J., Allen, M. B., Ismail-Zadeh, A. D., Flecker, R., Foland, K. A., & Simmons, M. D. (2005). Insights from the Talysh of Azerbaijan into the Paleogene evolution of the South Caspian region. Bulletin of the Geological Society of America, 117, 1513–1533.

Von Blanckenburg, F. (2006). The control mechanisms of erosion and weathering at basin scale from cosmogenic nuclides in river sediment. *Earth and Planetary Science Letters*, 242(3-4), 224–239.

- Walker, J. C., Hays, P. B., & Kasting, J. F. (1981). A negative feedback mechanism for the long-term stabilization of Earth's surface temperature. Journal of Geophysical Research, 86(C10), 9776–9782.
- Wang, J. G., Hu, X., Garzanti, E., Ji, W. Q., Liu, Z. C., Liu, X. C., & Wu, F. Y. (2017). Early cretaceous topographic growth of the Lhasaplano, Tibetan Plateau: Constraints from the Damxung conglomerate. *Journal of Geophysical Research: Solid Earth*, 122, 5748–5765. https://doi. org/10.1002/2017JB014278
- Wei, Y., Zhang, K., Garzione, C. N., Xu, Y., Song, B., & Ji, J. (2016). Low palaeoelevation of the northern Lhasa terrane during late Eocene: Fossil foraminifera and stable isotope evidence from the Gerze Basin. *Scientific Reports*, *6*, 27508.
- Willeit, M., Ganopolski, A., Calov, R., Robinson, A., & Maslin, M. (2015). The role of CO<sub>2</sub> decline for the onset of Northern Hemisphere glaciation. *Quaternary Science Reviews*, 119, 22–34.
- Willenbring, J. K., & Jerolmack, D. J. (2016). The null hypothesis: Globally steady rates of erosion, weathering fluxes and shelf sediment accumulation during Late Cenozoic mountain uplift and glaciation. *Terra Nova*, 28(1), 11–18.
- Willenbring, J. K., & von Blanckenburg, F. (2010). Long-term stability of global erosion rates and weathering during late-Cenozoic cooling. Nature, 465, 211–214.
- Wu, F. Y., Ji, W. Q., Wang, J. G., Liu, C. Z., Chung, S. L., & Clift, P. D. (2014). Zircon U–Pb and Hf isotopic constraints on the onset time of India-Asia collision. American Journal of Science, 314(2), 548–579.
- Xu, R. (1981). Vegetational changes in the past and the uplift of Qinghai-Xizang Plateau. In Proc. Symp. on Qinghai-Xizang (Tibet) Plateau, (Vol. 1, pp. 139–144). Beijing, China: Science Press.
- Yin, A., & Harrison, T. M. (2000). Geologic evolution of the Himalayan-Tibetan orogen. Annual Review of Earth and Planetary Sciences, 28, 211–280.
- Zachos, J., Pagani, M., Sloan, L., Thomas, E., & Billups, K. (2001). Trends, rhythms, and aberrations in global climate 65 Ma to present. *Science*, 292(5517), 686–693.
- Zachos, J. C., Dickens, G. R., & Zeebe, R. E. (2008). An early Cenozoic perspective on greenhouse warming and carbon-cycle dynamics. *Nature*, 451(7176), 279.
- Zeebe, R. (2012). LOSCAR: Long-term Ocean-atmosphere-Sediment CArbon cycle Reservoir Model v2.0.4. Geoscientific Model Development, 5, 149–166.
- Zhu, D. C., Wang, Q., Zhao, Z. D., Chung, S. L., Cawood, P. A., Niu, Y., et al. (2015). Magmatic record of India-Asia collision. Scientific Reports, 5, 14289.
- Ziegler, A. M., Scotese, C. R., McKerrow, K. S., Johnson, M. E., & Bambach, R. K. (1979). Paleozoic paleogeography. Annual Review of Earth and Planetary Sciences, 7(1), 473–502.